

Trinity College Dublin Coláiste na Tríonóide, Baile Átha Cliath The University of Dublin

INFORMATION BIAS IN THE GRAPHICS ACCOMPANYING NEWS ARTICLES

Cian Flynn

A dissertation submitted in partial fulfilment of the degree of MAI Computer Engineering

Supervisor: Professor Séamus Lawless

Submitted to the University of Dublin, Trinity College, May 2018

Declaration

I, Cian Flynn, declare that the following dissertation, except where otherwise stated, is entirely my own work; that it has not previously been submitted as an exercise for a degree, either in Trinity College Dublin, or in any other University; and that the library may lend or copy it or any part thereof on request.

Signature:

Date: 10th May, 2018

Summary

Bias in the mainstream news media has been studied for many years. In recent years, traditional mediums have given way to online and digital mediums, and with them new features are being presented to readers which could influence bias. This research aimed to determine if the level of bias perceived to be in a news article online could be impacted by an information graphic accompanying that article. The experiment involved participants reporting how biased they considered an article to be for three news stories, from three websites. Each story from each website was studied at three levels of distortion: with no graphic, with the original graphic, and with a tailored graphic.

The research found that an information graphic accompanying a news article will affect the level of bias perceived for the article as a whole, but that this effect was not found to be statistically significant, and no trend was found regarding what causes the bias level to increase or decrease. With a sample size of 124 participants, no statistically significant effect was detected. Closer examination of the data collected reveals findings which suggest further work would be worthwhile in this area.

The data collected also indicates strong tendencies among participants to avoid cognitive dissonance, and the presence of confirmation bias on the part of the participants. When participants were presented with a website they recognised, and which they previously had considered to be biased, they gave a significantly higher rating for bias in that article than for those who did not recognise the website. Additionally, those who recognised the website but did not previously consider it to be particularly biased, reported far lower levels of bias for that article than those who had not recognised it.

This research adds to an existing body of work in the area. Possible refinements to the study are proposed.

Acknowledgements

I would like to acknowledge the work of Professor Séamus Lawless, without whose initial guidance I would not have undertaken this research, and who has provided constant support and guidance throughout the project.

I would also like to acknowledge the extraordinary amount of advice and support given to me by Mr. Brendan Spillane. The motivation and guidance he provided throughout the research ensured that I could complete this research with the highest degree of academic rigour and integrity, which he promotes constantly.

Finally, I must acknowledge the support and help received from my family and friends over the course of this research project.

I thank them all for everything.

Cian Flynn

May 2018

Table of Contents

Dec	claration	ii
Sun	nmary	. iii
Ack	nowledgements	. iv
Tab	ole of Contents	v
Tab	ole of Figures	ix
Tab	ble of Tables	xi
1.	Introduction	1
1	1.1 Research Question	3
1	I.2 Hypothesis	3
1	L3 Overview of this Dissertation	4
2.	Existing Work	5
2	2.1 Bias	5
	2.1.1 Definitions of Bias	6
	2.1.2 Bias in Mainstream News Media	7
	2.1.3 Bias in the Presentation of News Online	8
	2.1.4 Information Graphics in News Media	8
	2.1.5 Pigs in the Procentation of Information Craphics	0
	2.1.5 bias in the Presentation of information Graphics	9
2	2.2 Example of Driving a Narrative with Visualisation	. 10
2 2	2.2 Example of Driving a Narrative with Visualisation	. 10 . 11
2 2 3.	2.1.5 Blas in the Presentation of Miormation Graphics 2.2 Example of Driving a Narrative with Visualisation 2.3 Conclusion Methodology	. 10 . 11 . 12
2 2 3. 3	2.1.5 Blas in the Presentation of Miormation Graphics 2.2 Example of Driving a Narrative with Visualisation 2.3 Conclusion Methodology 3.1 Websites	. 10 . 11 . 12 . 12
2 2 3. 3	2.1.5 Blas in the Presentation of Miormation Graphics 2.2 Example of Driving a Narrative with Visualisation 2.3 Conclusion Methodology 3.1 Websites 3.2 Articles	. 10 . 11 . 12 . 12 . 12
2 2 3. 3 3	 2.1.5 Blas in the Presentation of miorination Graphics. 2.2 Example of Driving a Narrative with Visualisation. 2.3 Conclusion. Methodology. 3.1 Websites. 3.2 Articles . 3.3 Distortions. 	.10 .11 .12 .12 .12 .12
2 2 3. 3 3	 2.1.5 Blas in the Presentation of mormation Graphics	.10 .11 .12 .12 .12 .12 .13
2 2 3. 3 3 3	 2.1.5 Blas in the Presentation of mormation Graphics. 2.2 Example of Driving a Narrative with Visualisation. 2.3 Conclusion. Methodology. 3.1 Websites. 3.2 Articles . 3.3 Distortions. 3.3.1 Creating D2 Graphics . 3.3.2 Website Branding. 	.10 .11 .12 .12 .12 .13 .13 .14
2 2 3. 3 3 3 3	 2.1.5 Blas in the Presentation of Mormation Graphics	.10 .11 .12 .12 .12 .12 .13 .13 .14 .15
2 2 3. 3 3 3 3 3	 2.1.5 Bas in the Presentation of mormation Graphics. 2.2 Example of Driving a Narrative with Visualisation. 2.3 Conclusion. Methodology. 3.1 Websites. 3.2 Articles 3.3 Distortions. 3.3.1 Creating D2 Graphics	.10 .11 .12 .12 .12 .13 .13 .13 .14 .15
2 2 3. 3 3 3 3 3	 2.1.5 Bias in the Presentation of mormation Graphics. 2.2 Example of Driving a Narrative with Visualisation. 2.3 Conclusion. Methodology. 3.1 Websites. 3.2 Articles . 3.3 Distortions. 3.3.1 Creating D2 Graphics . 3.3.2 Website Branding. 3.4.1 Phase 1: Participant Profile Questions. 3.4.2 Phase 2: Bias Rating Experiment. 	.10 .11 .12 .12 .12 .13 .13 .14 .15 .15 .16
2 2 3. 3 3 3 3 3	 2.1.5 Bas in the Presentation of Monnation Graphics. 2.2 Example of Driving a Narrative with Visualisation. 2.3 Conclusion. Methodology. 3.1 Websites. 3.2 Articles 3.3 Distortions. 3.3.1 Creating D2 Graphics . 3.3.2 Website Branding. 3.4.1 Phase 1: Participant Profile Questions. 3.4.2 Phase 2: Bias Rating Experiment. 3.4.3 Experiment Paths . 	.10 .11 .12 .12 .12 .13 .13 .13 .14 .15 .15 .16
2 2 3. 3 3 3 3 3	 2.1.5 Bas in the Presentation of information Graphics. 2.2 Example of Driving a Narrative with Visualisation. 2.3 Conclusion. Methodology. 3.1 Websites. 3.2 Articles . 3.3 Distortions. 3.3.1 Creating D2 Graphics	.10 .11 .12 .12 .12 .13 .13 .13 .13 .14 .15 .16 .16 .18
2 2 3. 3 3 3 3 3 3	 2.1.5 Bias in the Presentation of mornation Graphics. 2.2 Example of Driving a Narrative with Visualisation. 2.3 Conclusion. Methodology. 3.1 Websites. 3.2 Articles 3.3 Distortions. 3.3.1 Creating D2 Graphics . 3.3.2 Website Branding. 3.4 The Experiment. 3.4.1 Phase 1: Participant Profile Questions. 3.4.2 Phase 2: Bias Rating Experiment. 3.4.3 Experiment Paths. 3.4.4 Phase 3: Reflective Questions. 3.5 Bias Rating Scale. 	.10 .11 .12 .12 .12 .13 .13 .13 .14 .15 .15 .16 .16 .18 .19
2 2 3. 3 3 3 3 3 3 3 3	 2.1.5 bias in the Presentation of Miorination Graphics. 2.2 Example of Driving a Narrative with Visualisation. 2.3 Conclusion. Methodology. 3.1 Websites. 3.2 Articles 3.3 Distortions. 3.3.1 Creating D2 Graphics 3.3.2 Website Branding. 3.4 The Experiment. 3.4.1 Phase 1: Participant Profile Questions. 3.4.2 Phase 2: Bias Rating Experiment. 3.4.3 Experiment Paths 3.4.4 Phase 3: Reflective Questions. 3.5 Bias Rating Scale. 3.6 Participants 	.10 .11 .12 .12 .12 .13 .13 .13 .13 .14 .15 .15 .16 .16 .18 .19 .19

4	Experiment Apparatus	22
	4.1 Prolific Academic	22
	4.2 Begin Survey	23
	4.3 Participant Information	23
	4.4 Instruction Tasks	23
	4.5 Survey Questions	23
	4.6 Spike Question	24
	4.7 Bias Ratings	24
	4.8 Reflective Survey Questions	24
	4.9 Debrief	25
	4.10 Submission and Payment	25
	4.11 Conclusion	25
5	Data Analysis	26
	5.1 ANOVA Assumptions	26
	5.2 Kruskel-Wallis H Test Assumptions	27
	5.3 Kruskel-Wallis H Test Hypotheses	28
	5.3.1 The Null Hypothesis	28
	5.3.2 Alternative Hypothesis	29
	5.4 Kruskal-Wallis H Test Results	30
	5.5 Conclusion	30
6	Results	31
	6.1 Experiment Participants	31
	6.2 How to Read the Kruskel-Wallis H Test Results	32
	6.2.1 Sample breakdown of results for a Kruskel-Wallis H Test interpretation	33
	6.3 Bias Levels by Distortion	36
	6.3.1 W0A0 (D0/D1/D2) – The Daily Mail, CPI Inflation	36
	6.3.2 W0A1 (D0/D1/D2) – The Daily Mail, Bitcoin Prices	36
	6.3.3 W0A2 (D0/D1/D2) – The Daily Mail, Life Expectancy	37
	6.3.4 W1A0 (D0/D1/D2) – The BBC, CPI Inflation	37
	6.3.5 W1A1 (D0/D1/D2) – The BBC, Bitcoin Prices	38
	6.3.6 W1A2 (D0/D1/D2) – The BBC, Life Expectancy	38
	6.3.7 W2A0 (D0/D1/D2) – The Guardian, CPI Inflation	39
	6.3.8 W2A1 (D0/D1/D2) – The Guardian, Bitcoin Prices	39
	6.3.9 W2A2 (D0/D1/D2) – The Guardian, Life Expectancy	40
	6.3.10 Overview of Results	40
	6.4 Bias Levels by News Story	41

	6.4.1 W0D0 (A0/A1/A2) – The Daily Mail, no graphic	.41
	6.4.2 W0D1 (A0/A1/A2) – The Daily Mail, original graphic	41
	6.4.3 W0D2 (A0/A1/A2) – The Daily Mail, experiment graphic	42
	6.4.4 W1D0 (A0/A1/A2) – The BBC, no graphic	42
	6.4.5 W1D1 (A0/A1/A2) – The BBC, original graphic	43
	6.4.6 W1D2 (A0/A1/A2) – The BBC, experiment graphic	43
	6.4.7 W2D0 (A0/A1/A2) – The Guardian, no graphic	44
	6.4.8 W2D1 (A0/A1/A2) – The Guardian, original graphic	44
	6.4.9 W2D2 (A0/A1/A2) – The Guardian, experiment graphic	45
	6.4.10 Overview of Results	. 45
6	.5 Bias Levels by Website	46
	6.5.1 A0D0 (W0/W1/W2) – CPI Inflation, no graphics	46
	6.5.2 A0D1 (W0/W1/W2) – CPI Inflation, original graphics	46
	6.5.3 A0D2 (W0/W1/W2) – CPI Inflation, experiment graphics	47
	6.5.4 A1D0 (W0/W1/W2) – Bitcoin Prices, no graphics	47
	6.5.5 A1D1 (W0/W1/W2) – Bitcoin Prices, original graphics	48
	6.5.6 A1D2 (W0/W1/W2) – Bitcoin Prices, experiment graphics	48
	6.5.7 A2D0 (W0/W1/W2) – Life Expectancy, no graphics	49
	6.5.8 A2D1 (W0/W1/W2) – Life Expectancy, original graphics	49
	6.5.9 A2D2 (W0/W1/W2) – Life Expectancy, experiment graphics	. 50
	6.5.10 Overview of Results	. 50
6	.6 Summary of Results	51
6	.7 Conclusions	. 52
	6.7.1 Bias Rating Results	. 52
	6.7.2 Level of Significance	. 52
7.	Discussion of Results	. 54
7	.1 Bias Rating Results	. 54
	7.1.1 The Daily Mail	. 54
	7.1.2 The BBC	. 54
	7.1.3 The Guardian	. 55
	7.1.4 Website and Article Dimensions	. 55
	7.1.5 Impact of Distortions	. 55
	7.1.6 Impact of Graphics	. 56
7	.2 Hypothesis Results	. 56
7	.3 Bias Attributed to Publication	. 56
	7.3.1 Participant Recognition of Websites	.57

7.3.2 Participants' Opinions of the Websites' Bias	59	
7.3.3 Effect on Bias Ratings	60	
7.3.4 Results of Bias Testing	61	
7.3.5 Analysis of Results for bias attributed to publications	62	
7.4 Other External Factors	63	
7.5 Factors Contributing to Bias	63	
7.5.1 Responses	64	
7.5.2 Analysis of Responses	65	
7.6 Summary of Results	65	
8. Conclusions	67	
8.1 Future Work and Refinements	68	
8.2 Final Thoughts	69	
References	xii	
Appendix A1: Bias Ratings Data	xv	
Appendix A2: Survey Responses	xxi	
Appendix A3: Participant Information	xxix	
Appendix A4: News Articles and Distortions Used in the Experiment	xxx	
Appendix A5: Debrief Page	xxxix	
Appendix A6: Background Results	xl	
A6.1 Political Leaning vs Reported Bias, Daily Mail	xl	
A6.2 Political Leaning vs Reported Bias, BBC	xl	
A6.3 Political Leaning vs Reported Bias, Daily Mail	xli	
A6.4 Occupation vs Reported Bias	xli	
Appendix A7: Instruction Tasks	xlii	
Appendix A8: D1 and D2 Graphicsxliv		
Appendix A9: Chi-squared Tablesxlix		

Table of Figures

Figure 2.1 – Visualisation of population growth vs. population. Image source: Our World in Data 10
Figure 3.1 - Original graphic (D1), left. Especially created graphic (D2), right14
Figure 3.2 – Example of BBC website banner with explicit branding removed
Figure 3.3 - An illustration of the 3x3x3 experiment cube, showing the three levels of websites, articles, and distortions. Each shaded layer of the cube corresponds to a table below (Tables 3.1 – 3.3), which detail how each path traverses the cube
Figure 3.4 - The Visual Analogue Scale on which participants rated the perceived level of bias for each article
Figure 5.1 - Boxplot showing data collected for all twenty-seven combinations of WAD presented to participants. Significant outliers are seen represented by circles (more than one box-length), and asterisks (more than three box-lengths). Note that the numbers associated with the outliers are the unique identification numbers for those points in the dataset, and not the values of those points27
Figure 5.2 - Example illustration showing three different groups whose distribution is different, with distributions of the same shape (left), and different shape (right). Image source: Laerd Statistics 29
Figure 6.1 - The age distribution of participants32
Figure 6.2 - Number of participants by gender32
Figure 6.3 – Results of the Kruskal-Wallis Test for W0A036
Figure 6.4 - Results of the Kruskal-Wallis H Test for W0A1
Figure 6.5 - Results of the Kruskal-Wallis H Test for W0A1
Figure 6.6 - Results of the Kruskal-Wallis H Test for W1A037
Figure 6.7 - Results of the Kruskal-Wallis H Test for W1A1
Figure 6.8 - Results of the Kruskal-Wallis H Test for W1A2
Figure 6.9 - Results of the Kruskal-Wallis H Test for W2A0
Figure 6.10 - Results of the Kruskal-Wallis H Test for W2A1
Figure 6.11 - Results of the Kruskal-Wallis H Test for W2A240
Figure 6.12 - Results of the Kruskal-Wallis H Test for W0D041
Figure 6.13 - Results of the Kruskal-Wallis H Test for W0D141
Figure 6.14 - Results of the Kruskal-Wallis H Test for W0D2

Figure 6.15 - Results of the Kruskal-Wallis H Test for W1D0	42
Figure 6.16 - Results of the Kruskal-Wallis H Test for W1D1	43
Figure 6.17 - Results of the Kruskal-Wallis H Test for W1D2	43
Figure 6.18 - Results of the Kruskal-Wallis H Test for W2D0	44
Figure 6.19 - Results of the Kruskal-Wallis H Test for W2D1	44
Figure 6.20 - Results of the Kruskal-Wallis H Test for W2D2	45
Figure 6.21 - Results of the Kruskal-Wallis H Test for A0D0	46
Figure 6.22 - Results of the Kruskal-Wallis H Test for A0D1	46
Figure 6.23 - Results of the Kruskal-Wallis H Test for A0D2	47
Figure 6.24 - Results of the Kruskal-Wallis H Test for A1D0	47
Figure 6.25 - Results of the Kruskal-Wallis H Test for A1D1	48
Figure 6.26 - Results of the Kruskal-Wallis H Test for A1D2	48
Figure 6.27 - Results of the Kruskal-Wallis H Test for A2D0	49
Figure 6.28 - Results of the Kruskal-Wallis H Test for A2D1	49
Figure 6.29 - Results of the Kruskal-Wallis H Test for A2D2	50
Figure 6.30 – The p-values for each of the tests carried out on the bias ratings	52
Figure 7.1 – Number of participants claiming to recognise each website during the experiment, chosen from a multiple-choice list	58
Figure 7.2 – Number of participants correctly identifying a number of websites	59
Figure 7.3 – The number of participants who consider each news website to be particularly biased	d. 60
Figure 7.4 - Results of the Kruskal-Wallis H Test for the Daily Mail (W0)	61
Figure 7.5 - Results of the Kruskal-Wallis H Test for the BBC (W1)	61
Figure 7.6 - Results of the Kruskal-Wallis H Test for The Guardian (W2)	62
Figure 7.7 – Media regularly used by participants to consume news.	63
Figure 7.8 – Responses to the question, "When judging bias in news articles online, which of the following do you consider to be significant elements of bias?"	64

Table of Tables

1. Introduction

With the ever-increasing availability and consumption of news media online, it is increasingly important that we understand the factors which affect our interpretation of news online. Bias in the online news media industry has, many times in recent years, been used as an explanation for a rise in support for political extremes [1]. Recently it has been credited with influencing the outcome of US Presidential and local elections [2], and for driving public support for a referendum decision which will result in the United Kingdom leaving the European Union [3]. It is evident that there is widespread belief that bias in news media, and specifically online news media, can be used to drive major social, economic, and political changes [4]. Bias has, and will continue to be an issue with news media, and has received significant attention from researchers for at least the last half century [5], [6].

Due to increases in the ability of modern printing techniques and the desire to attract attention to a news article dealing with a complex subject or series of events, newspapers regularly commission accompanying artwork for news articles in the form of information graphics. This method of displaying information is suitable for conveying large volumes of data quickly and accurately [7]. For example, the share price of a company can be presented as a graph over time, or the number of new cars being purchased could be presented as a bar chart per quarter. Traditionally, print news media was constituted almost entirely of text; the cost of printing images, be they photos or graphics, was prohibitive for printing on a frequent basis. With the falling cost of printing through technological advances, and the increased electronic consumption of news [8], images, including information graphics, have become more prevalent in news journalism. With the increasing presence of such images, comes increasing power to frame a narrative. Although the impact of bias in news, and especially print news, has been regularly investigated in areas such as headlines, photographs, coverage, rhetoric, and agenda setting among others, there is no research on whether it can be introduced in graphics and the extent of such influence. If bias can be introduced in graphics, then it is important that this effect is explored and its extent understood.

Information graphics are used in order to rapidly convey large amounts of information in easily interpretable ways. However, by changing the format of the information from raw data or other information into an information graphic, the designer of the graphic must make some design decisions; the graphic is inherently an abstraction of the original data, and is not a presentation of the original data itself [7]. Insofar as the information is being manipulated for presentation, this begs the question of whether bias can be introduced into these graphics or not, adding to or removing emphasis from a certain narrative construed from the original raw data.

Manipulating how facts, or which facts are presented to an uninformed viewer is central to the introduction of bias into news media. Bias in news media has been studied for decades, beginning with traditional news media formats, while online news has become an area of

research focus more recently. To date, some work, discussed in Chapter 2 of this dissertation, has explored how bias can be introduced through the text of news articles, and also through the design of the websites which contain them [9]. No significant work exists concerning whether bias can be introduced through the use of information graphics.

There are numerous definitions for the word "bias". The Oxford English Dictionary [10] defines bias as:

"[mass noun] Inclination or prejudice for or against one person or group, especially in a way considered to be unfair."¹

This definition is very broad and is untested in academic research. For this research, a more specific definition is required, pertaining to the field of study. Looking at existing literature shows a wide variety of definitions used.

Waldman and Devitt [11] stated, as part of their research into media bias:

"Bias can be defined as any systematic slant favoring one candidate or ideology over another."

Gentzkow and Shapiro [12], discussing content bias stated:

"All the accounts are based on the same set of underlying facts. Yet by selective omission, choice of words, and varying credibility ascribed to the primary source, each conveys a radically different impression of what actually happened. The choice to slant information in this way is what we will mean in this paper by media bias."

Andon and Free [13] suggest that media bias can be broken down into three categories: gatekeeping bias, where publishers decide what will be published and what won't be, coverage bias, where one side of a story gets more coverage than another, and statement bias, where bias is introduced through the text used in the coverage.

Bennet [14] also focusses on the significance of gatekeeping bias, but for this study only those things which are reported can be studied, and so this type of bias will not form a major part of this study.

Eberl, Boomgarden and Wagner [15] also suggest that there are a multitude of type of media bias, and that their effects may differ.

Across all of these definitions of bias however, it is clear that bias is not a result of untruth, but rather the selective framing of truth in favour of a particular point of view.

In the second and third definitions, which are more specifically tailored to the discussion of bias in the media, the authors conclude that the bias must be introduced by choice or systematically. While the systematic slant mentioned by Waldman and Devitt does not

¹ https://en.oxforddictionaries.com/definition/bias

necessarily mean that the slant is knowingly introduced, it is important to remember that the area of bias in information graphics, which is the focus of this research, has not previously been extensively studied and is not well understood. It is plausible, if not likely, that bias which has been introduced to articles through the use of these information graphics may have been introduced inadvertently or unknowingly. Therefore, for this study bias is defined as:

"Deliberate or accidental slant by the journalist, editor or publication to distort reality".

In studying this particular aspect of perceived bias in online news media, covering a littleresearched area, this research should make a significant contribution to our understanding of that perceived bias in online news media.

1.1 Research Question

The purpose of this research project is to answer the question:

"To what extent do the information graphics accompanying a news article impact upon the perceived bias of the article?"

To this end, the following research objectives were formed:

- A ground-truth bias rating for each of the news articles
- Whether the information graphics accompanying those news articles can impact upon the perceived bias of the article
- If such graphics have an impact, the extent to which the design of the information graphics can impact upon the perceived bias of the article

A significant body of work already exists exploring bias in news media, and bias in graphics, but little work has been published about the perceived bias in the graphics accompanying news articles in online news media.

1.2 Hypothesis

This research aimed to determine whether bias can be introduced to an article through the use of information graphics. The research aimed to find evidence that bias could be introduced, and consequently the hypotheses are as follows.

The null hypothesis is:

(H₀) This research finds no evidence of bias being introduced to a news article through the use of an information graphic.

This is determined by finding the level of bias both with and without the information graphics and finding no statistically significant difference in the two levels.

The alternative hypothesis is:

(H_A) This research finds evidence of bias being introduced to a news article through the use of an information graphic.

This is determined by finding the level of bias both with and without the information graphics and finding a statistically significant difference in the two levels of bias.

1.3 Overview of this Dissertation

The remainder of this dissertation is structured as follows. The following chapter gives an overview of existing understanding related to bias in news media, both online and more generally, and bias in the creation and reproduction of information graphics.

Chapter 3 provides an explanation of the methodology employed in this research project, both in terms of the collection and the analysis of the data recorded.

Following this, Chapter 4 gives an overview of the experiment framework from a technical and practical perspective, **while Chapter 5** discusses the statistical methods used to make determinations on the data collected during the experiment.

Chapter 6 details the results of the statistical tests carried out on the data **and Chapter 7** provides a discussion on the results of the experiment. This includes discussion about the results presented in Chapter 6, and provides a reflection on the participant profile data also collected during the experiment.

Finally, Chapter 8 provides a conclusion to the dissertation, discussing both what has been learned and how this can contribute to future work in the area.

The full set of the data collected during experimentation, and the materials used in the collection of that data, can be found in the appendices at the end of this dissertation.

2. Existing Work

Bias in news media has been the subject of academic research for many decades, focussing both on specific aspects of the news media, for example the driving of a partisan agenda through bias [16]–[18], and on news media more generally [4], [17], [19].

This chapter will give an overview of existing work in the area of media bias, discussing the findings of that work, and why the research question is an important addition to the body of work. The findings of other researchers, discussed in this chapter, should also contextualise the design and implementation decisions, and results, of this experiment.

2.1 Bias

Bias in news media has been studied extensively since the 1970s, with research focussing mostly on the traditional mediums of print newspapers [11], [20] and television [6]. More recently research has begun to investigate the effects of bias in online news publications [8], [21]. The extent to which bias has been studied is no surprise given its apparent impact on news consumers [3], [22].

While most acknowledge the existence of bias, it can be difficult to describe. There is no doubt that while bias is only one dimension of credibility [23], there are many dimensions to bias, and by which bias could be measured [17].

Selection bias is the introduction of bias by choosing to cover some material and not to cover other material [24], [25]. For example, a newspaper editor could decide not to report at all on a story which an editor finds not to be of interest to their readers. Similarly, an editor could choose to omit certain aspects of a story for similar reasons. This could convey a significantly different narrative to an audience than if all aspects of the story were reported, while never actually diverging in any way from the truth, or having any inaccurate reporting. The bias in this case is introduced by what is omitted rather than by any manipulation of what is included.

Bias can also be introduced by the way in which content is presented or covered. This is called presentation bias. Research has found a link between the perceived bias of commentary depending on to whom that commentary is credited [16], [26], and also in how reported facts are contextualised [27]. The practice of selecting certain facts and presenting them as being more important or salient to a particular point in order to promote a particular point of view is known as framing [28], [29].

Partisan bias is the favouring of one party or person in news coverage of political stories. Often this presents itself in the form of selection bias, where a news agency is more likely to publish quotes by, or that are favourable to, a person or party which their readers support [30]. The causes for partisan bias are very often driven by a desire to please the audience to which the news agency is selling its output [24]. Agenda setting bias extends from this, where media outlets favour reporting viewpoints which agree with their underlying economic and political stance [31], [32].

Given that what constitutes bias is a matter of debate, there is also no generally accepted gold standard for the measurement of bias. Measurements of bias can only be obtained by people reporting bias which they experience. This is often in the form of a user survey, and it is clear that bias can be introduced both in the creation of content and also in the consumption of that content [33], [34]. That is to say that in the interaction between an article and consumer, the bias can be introduced by either the article or the consumer. As such, bias in a particular article could be attributed either to the article itself or its creator, or it could be attributed to the reader of that article, and must be regarded as, at best, an informed opinion rather than an empirical fact.

The level of bias perceived for a particular article is also an inherently subjective determination; each reader will have their own opinion. Measurements of bias can only be considered accurate when considered in the context that bias will have a relative definition for different people, and that the measurement can never be made in any absolute sense [15]. That is to say that the level of bias is intrinsically subjective, but this subjectivity can be accepted.

2.1.1 Definitions of Bias

There are numerous definitions of the word bias, including many which have been used specifically for the purposes of research into media bias.

As already discussed in Chapter 1, numerous researchers have used their own definitions for bias [11], [12], [17], each tailored to their specific areas of research. Related work in other areas which relate closely to bias, for example accuracy [35], have also presented their own definitions.

Across all of these definitions, it is clear that bias in media is the favouring of one particular side of a story, but not usually the intentional dissemination of false information. This is taken in the context of reputable news agencies, who uphold high standards of journalistic integrity. There are, of course, certain other media outlets who do in fact spread falsehoods, but these will not be considered in the scope of this research. This means that while an article may be perceived to be biased, and may favour one side of a story over the other, or drive a particular narrative, the facts reported in the article may still be exclusively true.

Some researchers contend that bias should only be considered so if it has been introduced intentionally, or by choice [12]. Others, meanwhile, have suggested that bias should be considered only if it is introduced systematically [11]; this could be interpreted to include unintentional bias which has been introduced as a result of an unrelated decision, for example an editorial decision on a newspaper's design.

Determining whether or not bias is present in the information graphics accompanying news articles presents a relatively new research area, it is not implausible that any bias which has been introduced to articles through the use of information graphics may have been introduced unintentionally. Therefore, for the purpose of this research, bias is defined as:

"Deliberate or accidental slant by the journalist, editor or publication to distort reality".

This definition was presented to all participants in the experiment before they undertook the experiment.

2.1.2 Bias in Mainstream News Media

While the majority of news articles published in the mainstream media are written to established journalistic standards, often an intrinsic bias can exist in some published news articles [18]. This can be perceived as being biased either in the way it is created, or by those who consume it. The bias can exist for many reasons, including commercial, economic, political and personal reasons.

Bias in the creation of news content can be introduced either in terms of how stories are covered, or in terms of what stories are selected to be covered by a particular media outlet [17], [24]. Even within a particular topic, the creator of the content can frame certain aspects of a story as being particularly important or salient, thus influencing a consumer's understanding of the facts, while never actually compromising the accuracy of the reporting of those facts [12], [27]–[29].

The perceived bias in the consumption of news media can be influenced significantly by a consumer's personal opinions about the topic or story being covered [33]. Furthermore, it has been found that consumers tend to favour news outlets whose editorial stance is aligned with their social, economic, and political views, and to avoid those with which they disagree, reducing cognitive dissonance [34].

Indeed, it has been found that news agencies play into this fact for commercial reasons. For news for which customers are unlikely to obtain an alternative, independent, third-party viewpoint, a news agency can actually bolster its reputation for accuracy by reporting news in a manner which its customers find agrees with their world-view [12]. News agencies may do this regardless of whether or not this is the fairest way to report the news. If consumers are likely to gain an alternative, independent account of the information from an impartial third-party, the same news agencies are less likely to distort the information they present [36]. As such, increased competition in the media marketplace tends to lead to lower levels of bias [12]. Furthermore, for stories on which there is a general consensus of public opinion, reporting tends to be relatively objective, whereas for stories which are contentious or for which there is divided public opinion, news agencies tend to segment the market, and promote more extreme editorial stances in their reporting [36].

The widespread availability of online news content has allowed for increased partisanship in the news industry [9], leading to the so-called "echo chamber" effect, whereby consumers can choose to expose themselves only to news stories which please them [4], [37].

Consumers will often do this subconsciously, as they often can be unaware of their own personal bias on a topic, while being very aware of someone else's with whom they disagree [38].

This in turn leads to consumer distrust in the media, as stories which displease consumers can be easily dismissed as biased or misleading, regardless of whether or not they actually are [39]. As a result, the vast majority of Americans, including 87% of Republicans [40], believe that the majority of mainstream news media is biased, one way or another [4], [41]. Although there is some evidence of measurable bias in the online publications of mainstream news media [19], there is strong research evidence to suggest that this is not as widespread as some may think [33], [42], [43].

It is important to note that research has also found that a journalist having a strong personal viewpoint on a particular subject will not necessarily cause their work to be biased or to convey bias [17], [44]. While it can be accepted that a person's own views will temper any decisions they make, including professional editorial decisions, professional journalistic standards will often help to mitigate any error arising from this judgement.

2.1.3 Bias in the Presentation of News Online

Bias in the news media has been extensively studied since the 1970s, and this work has typically focussed on highlighting differences between liberal and conservative news agencies or on biased coverage of a particular topic.

Compared to traditional mediums, relatively little research has been carried out into how bias may be present in the online presentation of news. Some existing research has looked at textual content and website design [9], [45]. It has previously been shown that bias exists with respect to vocabulary used in online news articles [21]. It has also been shown that the design of the website can have a significant impact on how biased its content is perceived to be, with certain features of website design affecting the perceived bias of the content more than others [46].

2.1.4 Information Graphics in News Media

Information graphics are used to convey large amounts of information in a format which can be interpreted quickly, and provide insights which the presentation of raw data alone could not achieve [7], [47], [48]. This should allow for increased understanding of a potentially large and complex set of information to be achieved by the reader, and relatively quickly. This means that vast amounts of information can be efficiently conveyed through this method of communication.

Images are also attention-grabbing, and so they can be used to attract a reader to read the article that they accompany [49].

Consequently, their use in media publications is frequent, indeed increasingly so [8], [50]. It follows that as their use increases, so too does their influence on communicating information from journalists to readers. This makes understanding the effects of this bias increasingly important.

2.1.5 Bias in the Presentation of Information Graphics

It has been noted by Munzner that: "no picture can communicate the truth, the whole truth, and nothing but the truth" [7]. It is clear that any representation of data in an informationgraphic is inherently subject to design decisions by its creator, and the resulting graphic is necessarily an abstraction of the data being represented and not just the data itself. Many of the factors which affect other forms of bias also affect bias in information graphics; bias can be introduced by consideration of what information is being communicated (selection bias), and also how it is being communicated (salience bias) [51].

It has been noted that even as information graphics in news media become more prevalent, little is understood about the balance between author-created elements of those graphics, and the data itself [52].

The processes for the creation of graphics also varies. The New York Times, for example, has a reasonably large, and multiple award-winning graphics department tasked with making these information graphics. Other publications such as The Economist and The Washington Post also spend significantly on graphic design. In many other publications less emphasis is put on graphic design, and so it is likely that information graphics which are presented are created on a more ad hoc basis. Given the standards of journalistic integrity which those who write the text of articles are held to, it should stand that the designers of graphics are held to the same standards, but as relatively little is known about the effects of the design process on readers' interpretation, this may not be the case.

Relatively little research has been carried out about how users extract information from information graphics. It has been shown that the design of the information graphics can affect how accurately and quickly readers can understand the information being conveyed [51], and also that the addition or removal of certain design elements can affect whether the information is perceived to be biased [53].

As information graphics accompany news articles with increasing frequency, it is increasingly important to understand how the use of these graphics can impact on the level of perceived bias in the article as a whole.

2.2 Example of Driving a Narrative with Visualisation

Figure 2.1 shows an example of an information graphic which presents the same information in two very different ways. The graphic visualises the population of the world, and also the growth rate of the population over the same period.



Figure 2.1 – Visualisation of population growth vs. population. Image source: Our World in Data²

Presenting both of these pieces of information related to the same thing on the same graphic is an eminently sensible design decision. However, if only one or the other of these pieces of information was presented, it would paint a very different picture.

A graphic with just the blue area visualising population growth could accompany an article about world population, and would signify a rapidly growing population. With this based in fact, and visually impressed on the reader, the article could make a claim about how this could be a huge problem for world resources, and seem very grounded in fact.

On the other hand, a graphic with just the line signifying the rate of population growth could accompany an article about world population, and would signify a rapidly slowing population growth. With this based in fact, and visually impressed on the reader, the article could make

² https://ourworldindata.org/world-population-growth

a claim about how the slowing population growth poses an existential threat to humankind, and this would seem very grounded in fact.

Considering either of those articles with the alternative graphic would expose a weakness in the case presented by either article, and shows the power of graphics to give credibility to the narrative in the text which they accompany.

In fact, for either case, the intentional selection of only certain elements of the graphic in order to bolster the claims made by the article shows a bias, which could massively influence the interpretation of the article by the reader.

However, it must also be considered that in Figure 2.1, the population is given as an absolute figure, whereas the population growth is given as a percentage of the population. This means that population is presented on a constant scale across the graphic, whereas population growth is presented on a varying scale. This shows how the scale can also be used to visually trick a reader into interpreting trends in data differently from how they actually are. The two elements of the graphic in Figure 2.1, are not directly comparable even though they are presented with each other.

2.3 Conclusion

Bias, and more specifically bias in the mainstream media has been studied extensively for decades. However, despite this, there has been almost no research carried out on bias introduced to media publications through the use of information graphics, and what effect this has on a reader's perception of the articles which the graphics accompany.

As graphics become ever more prevalent, particularly with the rise in online news readership, where their publication is no more expensive than text-based content, understanding how they affect our judgement of the articles they accompany, and whether they could be manipulated to favour one side of a story, or drive a particular narrative within that story, is more important than ever before. In a time when media bias is very much in the public spotlight, having been credited with driving major social, political, economic changes right around the world, understanding how bias can be introduced to news media online is crucial to understanding how this bias affects readers of these news publications.

This study will investigate whether bias is communicated by the graphical elements of news content which accompany articles in news websites, and attempt to determine what effect these information graphics may have on increasing or decreasing the level of perceived bias in the news article.

3. Methodology

Answering the questions posed by this research requires the collection and analysis of data related to how people perceive bias in online news media, specifically when they are viewing news articles accompanied by information graphics.

To do this, the experiment involved three stages:

- 1. Presenting participants with a survey to gather sample profile information
- 2. Then presenting participants with a series of images of webpages and asking them to rate how biased they considered the article on each of the webpages to be
- 3. The participants were then required to answer some follow up questions

The experiment was set up as a 3x3x3 design, Websites x Article x Distortions (WAD), using a Latin Cube in reduced form. This is a very simple form of Latin Hypercube Sampling, which is a statistical method for generating near random samples from a multidimensional set of parameters [54]. Participants were randomly assigned to one of nine paths through the 3x3x3 cube.

Participants were recruited to take part in the experiment through Prolific Academic [55]. The number of participants was determined using G*Power [56], in order to create a sufficient statistical power for the results to be meaningful.

Approval for this experiment was sought from, and granted by, the SCSS Research Ethics Committee, at the School of Computer Science and Statistic, Trinity College, Dublin.

3.1 Websites

The three websites selected were:

- W0. The Daily Mail [57]
- W1. The BBC [58]
- W2. The Guardian [59].

They were chosen to provide a cross-section of British news media. The Daily Mail is traditionally aligned with conservative and right-wing viewpoints, whereas The Guardian is traditionally aligned with liberal and left-wing viewpoints. As a public service broadcaster, the BBC is traditionally seen as an impartial reporter [60]. The study therefore uses the UK's mainstream media perceived as among the most right-wing, left-wing, and from the centre.

3.2 Articles

Articles were chosen from the Daily Mail, The Guardian, and the BBC, covering stories reported by all three news agencies on the same day, and which contained graphical

representations of data relating to the story. As such, any bias perceived to be present in the articles was not related to story selection to fit the experiment.

The three articles selected were on the following stories:

- A0. Consumer Price Index inflation, published on December 12th, 2017
- A1. The price of bitcoin, published on February 2nd, 2018
- A2. The rise in life-expectancy, published on September 27th, 2017

These three articles were chosen as they satisfied the criteria for inclusion. Importantly, they were not of excessive length which could have resulted in fatigue effects tempering the responses from participants. All three websites also reported very similarly in terms of data and sources, making them suitable for comparison.

3.3 Distortions

There are three levels of distortion for each website and story:

- D0. The article presented without any accompanying information graphics. This is to determine a ground truth bias rating for the text of the article. The articles were presented with all other features of their website design included, but with branding removed.
- D1. The article presented in its original form, including its original information graphic, but with branding removed. This is to determine the bias rating for the article as it was originally published.
- D2. The article presented with the original graphic replaced by a new graphic, specially designed for the purposes of this experiment.

3.3.1 Creating D2 Graphics

The D2 level of distortion required graphics to be created specifically for the purposes of this experiment. For example, Figure 3.1 shows two graphics, the original graphic (D1) from the BBC article about inflation (W1AO), and the newly introduced graphic (D2). The D2 graphic has been designed to suggest an exaggerated gap in the inflation of wages and the consumer price index, discussed in the article, while making this distortion of the figure not immediately obvious. The figures were sourced from the same source as the BBC graphic.

For each D2 graphic created, the figures have been sourced from the same sources as the original figures used, or else sources cited in the article. Some D2 graphics were entirely original creations, while others were modified version of D1 graphics, with significant design changes, in ways which are known to alter perception of information graphics.

All the distortions presented to participants are shown in Appendix A4. All the original D1 graphics, and the newly created D2 graphics are presented side-by-side in Appendix A8.

Inflation rate hits 3.1%



Figure 3.1 - Original graphic (D1), left. Especially created graphic (D2), right.

3.3.2 Website Branding

In the case of all three distortions, all branding, or anything which could identify the name of the website being used, was removed from the distorted versions of the websites. It is known that preconceptions about a newspaper's brand can influence a reader's trust in the content of that paper [45], and thus may impact their perception of bias. The branding was removed, as shown in Figure 3.2, in order to reduce the possibility of participants in the survey having preconceptions about the websites in question. This likelihood of this occurring was further reduced by limiting participation to US citizens living in the USA, and only using UK news websites. This was intended to increase the trust we can place in their responses being a considered and fair reflection of the content being shown. It is impossible to completely remove the possibility of participants recognising the websites, but by removing the branding, this concern is minimised as much as is possible, while without fundamentally altering the presentation of the website. It is important to avoid altering the presentation significantly, as this would unnecessarily introduce another variable to the experiment, which could not be independently measured; namely perceived bias possibly owing to the websites' design [46].



Figure 3.2 – Example of BBC website banner with explicit branding removed

3.4 The Experiment

As outlined in the opening section of this chapter, the experiment was split into three stages:

- 1. Participant Profile Questions
- 2. Bias Rating Experiment
- 3. Reflective Questions

The experiment was designed in this way so as to gather as much information as possible from each participant, not just about how biased they determined the articles with which they were presented to be, but also information which may give insight to how they reached that determination.

The three phases of the experiment are outlined in more detail below.

3.4.1 Phase 1: Participant Profile Questions

Rather than just collecting bias ratings from the participants, it was important to gain some context surrounding the responses provided by participants to the survey.

Prior to being presented with the websites for evaluation, participants were asked to provide the following information:

- Date of birth
- Gender
- Amount of time spent on the internet each day
- Highest level of education completed
- Expected income for 2018
- Occupation
- Political leaning
- Media accessed most often
- How many times do they check the news online each day

The data gathered from the responses to these questions is useful in contextualising the answers from the participants. If trends exist among socio-economic factors and the level of bias reported for a story, it would indicate an underlying effect which would require further attention and explanation. This can greatly add to the specificity of the findings.

3.4.2 Phase 2: Bias Rating Experiment

The bias rating part of the experiment involved presenting participants with an article from a website and asking the participant to report how biased they considered that article to be.

The participants rated bias on a slider scale from 0 to 100; 0 being not biased, and 100 being extremely biased.

The specific articles with which the participants were presented depended on to which path the participant had been assigned when they began the experiment. This system is outlined in the next section.

3.4.3 Experiment Paths

As outlined above, each participant was assigned to a path through the 3x3x3 cube. Each path ensured that each participant experienced each website, article, and distortion exactly once as they traversed the cube. The participants scored each website x article x distortion (WAD) combination presented to them on a slider scale from 0 to 100. The scale was anchored with the terms "unbiased" and "extremely biased".

The assignment of participants to paths through the experiment cube was random. In order to ensure that each WAD combination was considered by the same number of participants, and thus the responses recorded for each path would be of equal statistical merit, the same number of participants was assigned to each path. This is important, as many statistical methods, including ANOVA which is suitable for this type of experiment, require balance in the distribution of participants to groups.

This was achieved by setting a maximum number of participants for each path, and closing that path when the required number of participants had been assigned to it. Therefore, the requirement to have the same number of participants on each path did not affect the randomness of the allocation of paths.

There are twenty-seven possible WAD combinations in the 3x3x3 cube. By having nine separate paths through the cube, each participant was presented with just three combinations of article, website and distortion. The combinations shown to participants on each path are shown in Figure 3.1 and Tables 3.1–3.3 below.



Figure 3.3 - An illustration of the 3x3x3 experiment cube, showing the three levels of websites, articles, and distortions. Each shaded layer of the cube corresponds to a table below (Tables 3.1 – 3.3), which detail how each path traverses the cube.

Table 3.1 - A top-down view of the top (white) row in the 3x3x3 cube in Figure 3.1 above. The intersecting A0 / D0 cell is the same as the A0 / D0 cell in Figure 3.1 above. Participants who were assigned to path one would therefore experience Website 0 x Article 0 x Distortion 0.

	Article (D0)	Article &	Article &
		Original	Graphic (D2)
		Graphic (D1)	
Story 1 (A0)	Path 1	Path 2	Path 3
Story 2 (A1)	Path 4	Path 5	Path 6
Story 3 (A2)	Path 7	Path 8	Path 9

Table 3.2 - A top-down view of the middle (light grey) row in the 3x3x3 cube in Figure 3.1 above. The intersecting A0 / D0 cell is the same as the A0 / D0 cell in Figure 3.1 above. Participants who were assigned to path one would therefore experience Website 1 x Article 1 x Distortion 1

	Article (D0)	Article &	Article &
		Original	Graphic (D2)
		Graphic (D1)	
Story 1 (A0)	Path 9	Path 7	Path 8
Story 2 (A1)	Path 3	Path 1	Path 2
Story 3 (A2)	Path 6	Path 4	Path 5

Table 3.3 - A top-down view of the top (dark grey) row in the 3x3x3 cube in Figure 3.1 above. The intersecting A0 / D0 cell is the same as the A0 / D0 cell in Figure 3.1 above. Participants who were assigned to path one would therefore experience Website 2 x Article 2 x Distortion 2

	Article (D0)	Article &	Article &
		Original	Graphic (D2)
		Graphic (D1)	
Story 1 (A0)	Path 5	Path 6	Path 4
Story 2 (A1)	Path 8	Path 9	Path 7
Story 3 (A2)	Path 2	Path 3	Path 1

For example, a participant who was randomly assigned to Path 1 through the experiment, was presented with [W0, D0, A0], [W1, D1, A1] and [W2, D2, A2], and was asked to evaluate each of those three combinations.

Likewise, a participant who was randomly assigned to Path 2 through the experiment, was presented with [W0, D1, A0], [W1, D2, A1] and [W2 D0, A2], and was asked to evaluate each of those three combinations.

By having each participant evaluating just three of the twenty-seven possible permutations of WAD, the effects of fatigue are reduced. Also, the participant did not rate more than one distortion of a particular article or website, which makes any crossover effects impossible.

3.4.4 Phase 3: Reflective Questions

Having rated all three websites, the participants were asked the following:

- If they recognised any of the websites used in the experiment, from a provided list of news websites
- If they considered any of the websites to be particularly biased in general, from a provided list of news websites
- Which elements, from a multiple-choice list of supplied elements, they consider contribute to bias

This additional information was stored along with their other responses, and was completely anonymous.

In addition to this, participants were asked one 'spike' question. An image of an article about dogs was presented to each participant before the first article as part of this experiment. The participants were required to answer if the article was about dogs or cats, on the same slider scale as during the experiment, thus ensuring that the participants were paying attention to

the content being presented to them, and not just responding to the survey at random. This increases confidence that the responses to the survey were genuine and considered, and therefore that the conclusions drawn can be considered to reflect reality.

3.5 Bias Rating Scale

Bias was rated by participants on a visual analogue scale, which ranged from 0 to 100. This is shown in Figure 3.4. This type of scale was chosen as it allows participants to report precisely the level of bias they determine to be present in the article which they are rating.

An alternative type of scale, which has been used extensively in research in the past, is the Likert scale. On a Likert scale, the participants report their answer as an approximation which falls within a grouping of scores. This makes sense when evaluation of results is being carried out manually, as it makes for more simplified analysis. However, with automatic methods for data collection and analysis, the greater granularity of the responses made possible on a continuous scale was possible to utilise. It is for this reason that a continuous scale was utilised.





3.6 Participants

The participants for the experiment were sourced through the crowd sourcing marketplace Prolific Academic³[49].

Participation was limited to the United States of America. This was to help prevent the participants' personal opinion of articles presented to them affecting their perception of the articles' bias, as each of the news articles used for the experiment was sourced in media publications from the United Kingdom. With the branding removed from the UK websites, the likelihood of the American participants recognising the website, or having an opinion as to its bias, should be reduced [45]. If the participants did have their own strong opinion on the websites in question, it is likely that this could skew their interpretation of whether or not the article was biased [38].

The requisite number of participants was determined, using G*Power[56], in order to create a sufficient statistical power for the results to be meaningful. ANOVA analysis, discussed in

³ www.prolific.ac

Chapter 5, is generally suitable for this type of experiment, and so calculations were based on creating a statistical power for ANOVA. A statistical power of .45 or better is sufficient to draw meaningful conclusions from the research. In order to attain this statistical power, 121 participants would be required. However, in order to have an equal number of participants per path through the Latin Cube, 126 participants were required; 14 participants per path.

Having an α error probability of <0.05 means that we have a 95% chance of not making a Type 1 error.

Also, with a sample size of 126, there is a >0.95 Statistical Power, or a >95% probability, that we will detect an effect size of 0.45 when there is an effect to be detected, and not making a Type 2 error.

With 126 participants, the calculations are as shown in Table 3.4. The effect size, f, shows the effect size which is being sought. The power shows the probability of not making a Type 2 error for this effect size.

Input	Output
F Test	
ANOVA: Fixed effects, special, main effects and interactions Post hoc: Compute achieved power - given a sample size, and effect size	
• Effect size f = 0.45	Noncentrality Parameter = 25.515
 β / α ratio = 1 	• Critical F = 2.0674719
• Total Sample Size = 126	Denominator df = 117
 Numerator df = 8 (levels minus 1) 	• α err prob = 0.0444294
• Number of groups = 9	• β err prob = 0.0444294
	• Power = 0.9555706

 Table 3.4 - Table showing calculations of statistical power for this experiment, from G*Power.

3.7 Conclusion

The measurement of bias requires participants in the experiment to report their determination on the perceived level of bias in the article with which they are presented. As will be discussed in Chapter 5, in order to perform ANOVA analysis on the responses from the participants it is desirable to record a single continuous dependant variable, and one categorical independent variable. It was intended when designing the experiment that it should be suitable for this type of statistical analysis. Therefore, any other variables should be eliminated, and each participant should be asked to provide just a single piece of data; in this case, the level of bias in an article.

Participants are only asked to rate three different images of websites, thus reducing the possible effects of fatigue. Participants are also asked a spike question to ensure that they are paying attention before the experiment begins.

Use of Latin Hypercube Sampling provides a method of ensuring that participants are shown random images from the available selection, and also provides independence of observations between groups, another requirement for statistical analysis of the data collected.

The use of articles from UK websites, and American participants, along with removed branding, reduces the possible effects of bias arising from cognitive dissonance, by reducing the chances of the participants having a strong opinion on the material that they are rating. The presentation of the websites being kept as close as possible to their original format reduces the possible effects of bias being affected by the website design. It is important to mitigate these potential sources of error as much as is possible.

As statistical power of 0.45 is deemed sufficient to draw meaningful conclusions from this first research into the area; working with a higher statistical power would be economically prohibitive due to the number of participants required, while this should work to determine if there are useful results to be found from this experiment. According to Cohen [61], a power of 0.5 should detect an effect of medium magnitude. By using 0.45, the size which can be detected is even smaller. The calculations were based on the premise that ANOVA would be a suitable type of statistical analysis for the data collected in this experiment.

4. Experiment Apparatus

The survey for this experiment was deployed online, using a custom-built apparatus, on an Apache virtual machine (VM) in the School of Computer Science and Statistics at Trinity College, Dublin.

The apparatus is based on that built and used by Brendan Spillane for his web design experiments at the ADAPT Centre⁴, but used here with significant modification. The existing apparatus was designed to accommodate two-dimensional experiments up to a maximum size of a 9x9 two-dimensional Latin square. While significantly smaller than that size, this experiment required a framework capable of handling a three-dimensional cube. The apparatus' code, therefore, required heavy modification in order to be suitable for this experiment. This coding was completed and tested locally before being deployed to the VM for public use.

The apparatus is written primarily in the Python 2.7 programming language using the Django Framework. The HTML elements of the apparatus are styled using Twitter Bootstrap, and interactive elements of the design are controlled with JavaScript.

Responses by participants were stored on a MySQL database, running on the same VM as the front-end website. This was linked to the front-end using python-mysql.

The articles were modified for each required level of distortion with branding removed, using photo editing software. Each of these samples presented to the participants is shown in Appendix A4.

Each participant partaking in the survey progressed through the steps detailed in Sections 4.1 to 4.10.

4.1 Prolific Academic

The participants were sourced through the crowdsourcing marketplace Prolific Academic⁵.

Participation was limited to US citizens, living in the USA, and who were at least 18 years old.

Eligible participants saw this experiment advertised on Prolific Academic, chose to partake, and were brought to the start page of the experiment.

The experiment was advertised on Prolific Academic for four days, from April 27th to April 30th, 2018. Prolific Academic requires that participants are compensated at least £5 for an hour of participation in surveys. This experiment took approximately 10 minutes, and so each participant received compensation of £0.85 for their time.

⁴ https://www.adaptcentre.ie/

⁵ www.prolific.ac

4.2 Begin Survey

Participants who chose to partake were first brought to an introduction page where they were provided with detailed information about the experiment. They were then given the option to partake in the experiment or return to Prolific Academic. This page, and all subsequent pages were part of the website running on the VM at Trinity College, Dublin. At this point, participants clicked to begin their participation in the experiment.

4.3 Participant Information

As per Trinity Research Ethics, participants were presented with information about the experiment, and were then asked to provide their consent that their responses could be recorded and used in the analysis of this experiment. This was a prerequisite for participation in the experiment.

The full content of the participant information and consent pages are show in Appendix A3.

4.4 Instruction Tasks

Having provided their consent in the previous section, participants clicked the button to begin their participation in the experiment, and in so doing, were randomly assigned to one of the nine paths through the experiment, as described in Chapter 3, Figure 3.1.

They were then presented with instruction tasks; they were presented with pages which looked identical to the experiment proper, but were simply used to allow the participants to familiarise themselves with the experiment apparatus and the interface. Popup windows provided information to the user on how to partake in the experiment, and the users were requested to rate the bias of two sample images, as they would do later on in the experiment proper. These instruction pages are shown in Appendix A7.

4.5 Survey Questions

Before rating any articles as part of the experiment, participants were required to fill in a short survey. The results of this survey provided background information on the participants, allowing for contextualisation and greater understanding of their bias ratings.

The users were asked to provide the information detailed in Section 3.4.1.

This information provides information about the participants' socio-economic and political background. Though not directly contributing to the findings of the experiment, this information could be used to explain or contextualise the responses recorded.

4.6 Spike Question

Having completed the background survey, users were presented with a spike question to ensure diligence to the task and make sure they were not simply clicking though the experiment. This article is not part of the experiment data that was assessed and is simply designed to measure participants' attention to the task. To this end, participants were shown an article about dogs. The slider bar at the bottom of the page then asked them to indicate whether the article was about dogs or cats. Asking this simple and, more importantly, objective question for which there is a right and a wrong answer goes some way to ensuring that the participants were in fact paying attention to the experiment, and were giving considered responses. Any participant who answered the question incorrectly could be considered to have provided generally erroneous results, and their responses could be removed from consideration.

4.7 Bias Ratings

Participants then partook in the main part of the experiment, namely rating the bias they perceived to be present in each WAD combination. Each participant rated three WAD combinations, on a slider scale from 0 to 100, 0 being unbiased and 100 being extremely biased, as shown in Figure 3.4. The specific articles which were presented to the participants depended on which path they had been assigned through the experiment (see Figure 3.1).

After rating all three articles, the participants were then presented with all three again, with the opportunity to re-evaluate their ratings by comparing all three simultaneously.

Once satisfied with the rating which they had given for each, the participants were then once again presented with all three articles side-by-side, without seeing their previous rating, and asked to select the article they considered to be the most biased.

Asking for repeated affirmation of their decisions regarding the bias levels they perceived to be present in the articles increases confidence that the responses provided by the participants had been fully considered, and can provide meaningful information from which to draw conclusions.

4.8 Reflective Survey Questions

Having rated, and re-rated, the articles for their perceived level of bias, the participants were again asked a number of survey questions, reflecting on the experiment. Each participant was asked:

- If they recognised any of the websites used in the experiment, from a provided list of news websites
- If they considered any of the websites to be particularly biased in general, from a provided list of news websites

• Which elements, from a multiple-choice list of supplied elements, they consider contribute to bias

This information is used to provide context to the answers provided. For example, if a user said that they had recognised some of the websites used, we can no longer assume that their responses are entirely free of their own bias, from opinions they already had about the publication. Like the pre-experiment survey, described in Section 4.4, the answers to these questions do not directly contribute to the outcome of the experiment, but can be used to contextualise the responses recorded.

4.9 Debrief

Having completed the entire experiment, participants were presented with a debrief page, thanking them for their participation and providing them with contact information for the researchers, whom they could contact with any questions arising from the experiment. The full content of this page is presented in Appendix A6.

4.10 Submission and Payment

Having read the debrief page, participants were prompted to submit their responses to be included in the research. Upon clicking the button to do this, their answers were recorded on a MySQL database, which was hosted on the same Apache VM as the rest of the experiment apparatus.

The participants were then redirected to the Prolific Academic website, where they received payment as compensation for their time in completing the experiment.

4.11 Conclusion

The apparatus used in this experiment had previously been employed in two-dimensional experiments at the ADAPT Centre, but required heavy adaptation for use in this three-dimensional experiment.

The number of participants required to participate in the experiment was calculated using G*Power, as discussed in Chapter 3. The design of this experiment ensured that participants were always fully aware of what information they were expected to provide, and how to provide it. The design also ensured that responses were well considered, and provided context to a user's response decisions.
5. Data Analysis

Determining whether or not any bias exists in the data collected requires statistical analysis of the data. Originally, it was envisaged that analysis for this experiment would be achieved using one-way ANOVA, or Analysis of Variance tests, and the experiment was designed accordingly. On examination, it transpired that the data collected did not satisfy the assumptions for ANOVA as there were a number of significant outliers in the bias ratings, and so the Kruskel-Wallis H Test was performed instead. All tests were run using SPSS 25.

5.1 ANOVA Assumptions

There are six assumptions which an experiment and its resulting dataset must satisfy for ANOVA to be considered an appropriate test for that data:

- 1. There must be just one dependent variable, which must be continuous. In this case, that variable is the reported bias, which was reported on a continuous slider scale from 0 to 100 by the participants.
- 2. There must be one independent variable, which is categorical. In this case, that independent variable is one of either the Website, the Article, or the Distortion, depending on which dimension of the experiment is being examined at a particular time. Only one is examined at a time, which means that this assumption is satisfied.
- 3. There must be independence of observations. This is achieved by the Latin Hypercube Sampling experiment design, discussed in Chapter 3. This design ensures that the way that participants traverse the Latin Cube design guarantees independence of observations, by maintaining each randomly assigned group on separate paths through the cube.
- 4. There should be no significant outliers in the groups.
- 5. The dependent variable should be approximately normally distributed.
- 6. The variances of each variable should be approximately equal.

Assumptions 1, 2 and 3 are satisfied by the experiment design.

Assumptions 4, 5 and 6 depend entirely on the responses provided to the experiment by the participants. While it can often be expected that the responses will satisfy these assumptions, the data set must be tested to confirm that this is the case before an ANOVA analysis can be considered appropriate for the dataset.

Having collected all of the data for this experiment, the data failed Assumption 4; there were a small number of significant outliers, see Figure 5.1.



Figure 5.1 - Boxplot showing data collected for all twenty-seven combinations of WAD presented to participants. Significant outliers are seen represented by circles (more than one box-length), and asterisks (more than three box-lengths). Note that the numbers associated with the outliers are the unique identification numbers for those points in the dataset, and not the values of those points.

As there was no reason to reject these outlying data-points as being erroneous, it was necessary to include these in the statistical analysis. Consequently, ANOVA was unsuitable for analysis in this experiment, and an alternative, the Kruskel-Wallis H Test was used instead.

5.2 Kruskel-Wallis H Test Assumptions

Having rejected ANOVA as a suitable statistical test for use on this dataset, the Kruskel-Wallis H Test was chosen as a suitable alternative. This test, like ANOVA, is suitable for determining the relative differences between groups on a scale, which for this experiment is the bias rating. Importantly, the Kruskel-Wallis H Test is suitable for analysis on datasets with outliers, which is a necessity for the data collected in this experiment, as significant outliers exist.

For the Kruskel-Wallis H Test, the data must satisfy four assumptions, which are similar to the ANOVA assumptions, in order for this test to be suitable:

- There must be just one dependent variable, which can be either continuous ordinal. In this case, that variable is the reported bias, which was reported on a continuous slider scale from 0 to 100 by the participants.
- 2. There must be one independent variable, which is categorical. In this case, that independent variable is one of either the Website, the Article, or the Distortion, depending on which dimension of the experiment is being examined at a particular time. Only one is examined at a time, which means that this assumption is satisfied.

- 3. There must be independence of observations. This is satisfied by the experiment design as discussed in section 5.1.
- 4. A determination must be made of whether the distribution of scores of the dependent variable has the same or different shape for each of the independent variables, as shown in Figure 5.2. The test is suitable whether or not the shape is the same, but the alternative hypothesis is different depending on this determination. This determination is made for the data each time the test is run, which for this experiment is twenty-seven times over the three dimensions of the experiment cube.

As the data collected from the participants in this experiment satisfies these assumptions, the Kruskel-Wallis H Test is a suitable statistical method.

5.3 Kruskel-Wallis H Test Hypotheses

The research question which this experiment aims to answer is, "To what extent do the information graphics accompanying a news article impact upon the perceived bias of the article?". This requires determining if the perceived level of bias changes when a graphic is introduced, and if so what the nature of that change is. This is achieved by comparing the bias ratings at the three levels of distortion and for the three websites and news stories as discussed in Chapter 3.

The Kruskel-Wallis H Test can be used to determine if any differences observed in the collected data are statistically significant. The hypotheses which the Kruskel-Wallis H Test tests for are very similar to the hypotheses for ANOVA, and so it provides a suitable alternative test for this experiment. As this is a parametric test, unlike ANOVA, this test is suitable for datasets with significant outliers, which is a necessity for analysis of the data collected in this experiment.

The responses are grouped by their WAD combinations from the Latin Cube outlined in Chapter 3. The Kruskel-Wallis H Test is performed on each level of WAD, for each dimension of the cube. This means that twenty-seven tests are carried out on the experiment cube in total. The outcome of each test is either the rejection or retention of the null hypothesis, and in the case of rejection, the acceptance of the alternative hypothesis.

5.3.1 The Null Hypothesis

The null hypothesis (H₀), which is assumed, in a Kruskel-Wallis H Test is:

H_0 : The distribution of scores for the groups are equal.

For this experiment, the scores are the bias ratings by participants, and the groups are the combinations of WAD.

As it is highly improbable that a set of data will contain identical data in different independent groups, a determination must be made for what is a statistically significant difference in order

to reject the null hypothesis. For Kruskel-Wallis H Tests, this is determined by p-values. As is standard practice, a p-value of 0.05 was used to test for significance in this experiment.

5.3.2 Alternative Hypothesis

Alternatively, if the relative levels of bias between groups is found to be statistically significant, the null hypothesis must be rejected. This implies that the alternative hypothesis (H_1) may be accepted. Consequently, we determine that the averages of the populations are, in fact, different, and we can say with 95% certainty that the levels of bias are different in at least one of the three groups.



*Figure 5.2 - Example illustration showing three different groups whose distribution is different, with distributions of the same shape (left), and different shape (right). Image source: Laerd Statistics*⁶

Depending on whether the shapes of the distributions are equal, as per the example in Figure 5.2, determined in Assumption 4, the alternative hypothesis can be described in different ways. There is no generally accepted numerically rigorous approach to determining whether the shapes of the distributions are equal or not, and so this determination is made visually. For this experiment, that determination was made using box-plots of the data for each group. This is a standard approach.

If Assumption 4 is violated, and the shapes of the distributions are not similar then the alternative hypothesis states:

H_1 : The mean ranks for the groups are not equal.

This means that there is evidence that the level of bias reported at least one of the groups is statistically significantly different from the others. This means that bias has been found to exist. The precise extent to which the groups differ is found by comparing the mean ranks of the groups.

⁶ https://statistics.laerd.com/

Rejecting the null hypothesis, and consequently accepting the alternative hypothesis in all cases means that a statistically significant change in the levels of perceived bias has been detected in this experiment.

5.4 Kruskal-Wallis H Test Results

As mentioned in the previous section, whether or not to accept the null hypothesis is based on *p*-values. P-values become more accurate the larger the population size, but are considered appropriate for use once the population sample size is greater than five. The increase in accuracy is asymptotic, meaning that as the sample size increases, the rate of increase in the accuracy of the results reduces.

For this experiment, a *p*-value of <0.05 is considered to be statistically significant, as is standard practice. That is to say, a test result with a p-value of <0.05 will be sufficient to reject the null-hypothesis.

The p-value is calculated from the H-Statistic, which is the result of the experiment, and the number of degrees of freedom, which is the number of groups minus one. The H-Statistic approximately follows an x^2 distribution for the relevant number of degrees of freedom. For all tests of bias in this experiment, the number of degrees of freedom is two, as there are three groups in each of the three dimensions of the experiment cube.

5.5 Conclusion

In order to answer the research question proposed for this research, first a determination must be made as to whether or not the levels of bias are changed significantly by the introduction of graphics to news articles. The experiment was set up so that it would pass the first three assumptions necessary to undertake an ANOVA. However, the resulting data provided by the participants meant that there were too many outliers in the data and as a result it failed Assumption 4, rendering the dataset unsuitable for ANOVA analysis.

The Kruskel-Wallis H Test provides a suitable alternative to ANOVA. The test is suitable for determining a difference between the results reported by each group, but can do so regardless of whether or not there are outliers in the groups, or the distributions of responses are balanced. It was therefore used for analysis in this experiment.

The nature of any change found between the perceived levels of bias for each group, is determined by comparing either the mean rank or the median of the groups, depending on whether or not the distributions for each group had the same shape.

Accepting or rejecting the Kruskal-Wallis H Test's null hypothesis for these tests determined whether the level of bias perceived in articles was found to be affected by the use of information graphics accompanying those articles, answering the research question.

6. Results

This chapter details the results, following statistical analysis, of the bias ratings provided by participants in the experiment. The full data collected and used in this analysis is presented in Appendix A1.

The results shown below for each Kruskal-Wallis H Test carried out, comprise:

- A box plot, showing the reported levels of bias for each independent group being tested. The horizontal black line through the boxes shows the mean rank for each group.
- A table showing the mean value, and standard deviation for each independent group being tested. The mean value is on a scale from 0 to 100, as this is the range of responses allowed by the experiment apparatus.
- The decision on whether to accept or reject the null hypothesis, and the following key statistics drawn from the results
 - $\circ~$ The H Test Statistic, presented as Chi-squared, and represented by X²(N-1), where N is the number of groups being tested.
 - The p-value of the results, used to accept or reject the null hypothesis. A p-value of <0.05 is considered sufficient to reject the null hypothesis.

A full summary of these results is provided in Section 6.6. These results, including the visualisations, were generated using SPSS 25, as noted in Chapter 5.

6.1 Experiment Participants

The experiment was run, and all data collected over four days from April 27th to April 30th, 2018. 124 complete responses were gathered and considered for statistical analysis. A breakdown of the participants' demographics is shown in Figure 6.1 and 6.2. The vast majority of participants were aged between 18 and 40 years old. 70 of the participants, or 56%, identified themselves as being male. 52 of the participants, or 42%, identified as female.



Figure 6.1 - The age distribution of participants



Figure 6.2 - Number of participants by gender

A full breakdown of all responses to the experiment are shown in Appendix A1 and Appendix A2.

6.2 How to Read the Kruskel-Wallis H Test Results

This section is designed to show how the results presented in this chapter should be interpreted.

The determination on if the groups had a distribution of equal shape was made on the shape of the box plots for those groups. This decision is not dependent on how high or low the distribution is grouped on the vertical axis, but only on its height and the location of the mean rank within it, represented by the horizontal black line.

Regardless of whether or not the distributions were of equal shape, the Kruskel-Wallis H Test tests for the relative locations of the distribution on the vertical axis. The resulting score, the

H-statistic, is in fact a Chi-squared score for the test. As such, it is presented as X²(N-1), for N groups, as this is a more familiar statistic to many.

One of the results for this experiment is explained in detail in Section 6.2.1. All other results may be interpreted similarly.





Bias Reported			
Image Being Tested	Mean	N	Std. Deviation
W0A0D0.jpg	38.92	13	30.404
W0A1D0.jpg	45.88	17	28.295
W0A2D0.jpg	16.50	12	10.775
Total	35.33	42	27.716



The distributions of bias for each distortion were not similar for all groups. The values of bias reported for each article differed significantly from one another, $X^2(2) = 7.393$, p = 0.025. The null hypothesis is <u>rejected</u>.

6.2.1.1 Boxplots

The boxplot in the top left of figure 6.12 shows the distributions for the three groups being tested. In this case they are the groups of users who experienced W0A0D0, W0A1D0, and W0A2D0. It can be seen that the shapes of the three distributions of bias rating results are different from each other. The leftmost box has the largest distribution, and the rightmost box has the smallest distribution. This can also be confirmed with reference to the standard deviation, shown in the table on the right. It can also be seen that the median, the horizontal black line through each box plot, cuts the three boxes into different proportions. This is another indication that the distributions are not equal.

6.2.1.2 H-Statistic

To make a determination on whether or not to reject the null hypothesis for each test, the H statistic is calculated by running the Kruskal-Wallis H Test in SPSS 25. This is calculated by assigning a "rank" to each data point (bias rating), from 1 to n, beginning with the lowest bias rating, and working upward from there. A mean value is then found for the ranks which have been assigned to each group. This is called the mean rank for that group.

For example, the mean rank for the sample data set shown in Table 6.1 would be calculated as follows. Note, the data in Table 6.1 has been generated purely for explanation purposes, and is not real data from the experiment.

W0A0D0 Bias Ratings	W0A0D1 Bias Ratings	W0A0D2 Bias Ratings
15	45	18
32	63	22
43	42	46
17	19	82
14	32	53
28	61	37
53	25	46
26	36	57

Table 6.1 - Sample dataset with three groups of bias ratings.

1. The bias ratings are put in an ordered list and ranked from smallest to highest.

W0A0D0		W0A0D1	V0A0D1 W0A0D2			
Bias Rating	Rank	Bias Rating	Rank	Bias Rating	Rank	
15	2	45	15	18	4	
32	9	63	23	22	6	
43	13	42	12	46	16	
17	3	19	5	82	24	
14	1	56	20	54	19	
28	8	61	22	37	11	
53	18	48	17	44	14	
26	7	36	10	57	21	

Table 6.2- Sample dataset with three groups of bias ratings with ranks

2. The mean rank of each group is calculated. This is the arithmetic mean of the ranks assigned to that group.

Mean Rank for Group 1: (2 + 9 + 13 + 3 + 1 + 8 + 18 + 7)/8 = 61/8 = 7.625Mean Rank for Group 2: (15 + 23 + 12 + 5 + 20 + 22 + 17 + 10)/8 = 124/8 = 15.5Mean Rank for Group 3: (4 + 6 + 16 + 24 + 19 + 11 + 14 + 21)/8 = 115/8 = 14.375

3. The Kruskal-Wallis H Test then makes a comparison based on these mean ranks, rather than a comparison of the bias rating values themselves. This is the most suitable comparison of the distributions, as it takes into account the different distributions for each group.

The H-statistic is calculated using equation 1.

$$H = \left[\frac{12}{n(n+1)}\sum_{j=1}^{c}\frac{T_{j}^{2}}{n_{j}}\right] - 3(n+1)$$
(1)

where, n = sum of sample sizes for all samples, c = number of samples, T_j = sum of ranks of the jth sample, and n_j = size of the jth sample.

For the sample data in Table 6.1, this would be calculated, using Equation 1 as follows.

$$H = \left[\frac{12}{n(n+1)} \sum_{j=1}^{c} \frac{T_j^2}{n_j}\right] - 3(n+1)$$

$$H = \left[\frac{12}{24(24+1)} \left(\frac{61^2}{8} + \frac{124^2}{8} + \frac{115^2}{8}\right)\right] - 3(24+1)$$

$$H = \left[\frac{12}{600} \times \frac{16161}{4}\right] - 75$$

$$H = 5.805$$

The H-statistic, which is the result of the Kruskal-Wallis H Test, is often reported in the more familiar form of a chi-squared score. For this sample breakdown, the H-statistic is $H = X^2(2) = 5.805$, indicating that the experiment has two degrees of freedom. Higher H-statistics indicate a greater variance between the bias levels reported for each group. Similarly, lower H-statistics indicate a smaller variance between the bias levels reported for each group.

6.2.1.3 Hypothesis Testing and p-values

Using this H-statistic, and knowing the number of degrees of freedom, the *p*-value can be found, either by calculation or using tables for the Chi-squared distribution.

Using the Chi-square tables, as shown in Appendix A9, it can be seen that for a H-statistic/Chisquared score of 5.805, with two degrees of freedom, the *p*-value is in the range 0.10 > p > 0.05. As is standard practice, a *p*-value of <0.05 has been used in this experiment to determine significance. Having a *p*-value below 0.05 indicates that the null hypothesis can be rejected. For this example, *p* > 0.05, and therefore the null hypothesis is retained. This indicates that the difference between the bias reported for the groups is not statistically significant.

6.3 Bias Levels by Distortion

This section focusses on the outcome of the Kruskal-Wallis H Test for each combination of website and article. This directly compares the effects of distorting the graphic presented to users, without consideration of the website or article chosen.

For brevity, only the significant results are explored in detail.

6.3.1 W0A0 (D0/D1/D2) – The Daily Mail, CPI Inflation



Bias Reported		•	
Image Being Tested	Mean	N	Std. Deviation
W0A0D0.jpg	38.92	13	30.404
W0A0D1.jpg	53.07	14	34.582
W0A0D2.jpg	47.00	14	22.185
Total	46.51	41	29.322

Figure 6.3 – Results of the Kruskal-Wallis Test for WOAO

The distributions of bias for each distortion were not similar for all groups. The values of bias reported for each distortion differed from one another, but the differences were not statistically significant, $H = X^2(2) = 1.451$, p = 0.484. The null hypothesis is retained.

6.3.2 W0A1 (D0/D1/D2) - The Daily Mail, Bitcoin Prices



Bias Reported			
Image Being Tested	Mean	N	Std. Deviation
W0A1D0.jpg	45.88	17	28.295
W0A1D1.jpg	26.44	16	32.737
W0A1D2.jpg	32.50	14	32.938
Total	35.28	47	31.712



The distributions of bias for each distortion were not similar for all groups. The values of bias reported for each distortion differed from one another, but the differences were not statistically significant, $H = X^2(2) = 4.382$, p = 0.112. The null hypothesis is retained.

6.3.3 W0A2 (D0/D1/D2) – The Daily Mail, Life Expectancy



Bias Reported			
Image Being Tested	Mean	N	Std. Deviation
W0A2D0.jpg	16.5000	12	10.77455
W0A2D1.jpg	27.6667	12	31.93839
W0A2D2.jpg	36.1667	12	38.36389
Total	26.7778	36	29.77162

Figure 6.5 - Results of the Kruskal-Wallis H Test for WOA1

The distributions of bias for each distortion were not similar for all groups. The values of bias reported for each distortion differed from one another, but the differences were not statistically significant, $H = X^2(2) = 0.164$, p = 0.921. The null hypothesis is retained.

6.3.4 W1A0 (D0/D1/D2) - The BBC, CPI Inflation



Bias Reported			
Image Being Tested	Mean	Ν	Std. Deviation
W1A0D0.jpg	43.58	12	38.242
W1A0D1.jpg	28.08	12	25.040
W1A0D2.jpg	36.17	12	25.665
Total	35.94	36	30.082

Figure 6.6 - Results of the Kruskal-Wallis H Test for W1A0

The distributions of bias for each distortion were not similar for all groups. The values of bias reported for each distortion differed from one another, but the differences were not statistically significant, $H = X^2(2) = 1.180$, p = 0.554. The null hypothesis is retained.

6.3.5 W1A1 (D0/D1/D2) – The BBC, Bitcoin Prices



Bias Reported			
Image Being Tested	Mean	Ν	Std. Deviation
W1A1D0.jpg	27.29	14	30.040
W1A1D1.jpg	37.69	13	34.565
W1A1D2.jpg	43.93	14	29.531
Total	36.27	41	31.378

Figure 6.7 - Results of the Kruskal-Wallis H Test for W1A1

The distributions of bias for each distortion were not similar for all groups. The values of bias reported for each distortion differed from one another, but the differences were not statistically significant, $H = X^2(2) = 2.593$, p = 0.274. The null hypothesis is retained.

6.3.6 W1A2 (D0/D1/D2) – The BBC, Life Expectancy



Bias Reported			
Image Being Tested	Mean	Ν	Std. Deviation
W1A2D0.jpg	36.86	14	29.294
W1A2D1.jpg	49.12	17	29.801
W1A2D2.jpg	37.69	16	33.738
Total	41.57	47	30.913

Figure 6.8 - Results of the Kruskal-Wallis H Test for W1A2

The distributions of bias for each distortion were not similar for all groups. The values of bias reported for each distortion differed from one another, but the differences were not statistically significant, $H = X^2(2) = 1.983$, p = 0.371. The null hypothesis is retained.

6.3.7 W2A0 (D0/D1/D2) – The Guardian, CPI Inflation



Bias Reported			
Image Being Tested	Mean	Ν	Std. Deviation
W2A0D0.jpg	26.25	16	31.118
W2A0D1.jpg	26.21	14	28.631
W2A0D2.jpg	27.94	17	25.445
Total	26.85	47	27.808

The distributions of bias for each distortion were not similar for all groups. The values of bias reported for each distortion differed from one another, but the differences were not statistically significant, $H = X^2(2) = 1.036$, p = 0.596. The null hypothesis is retained.

6.3.8 W2A1 (D0/D1/D2) – The Guardian, Bitcoin Prices



Bias Reported			
Image Being Tested	Mean	Ν	Std. Deviation
W2A1D0.jpg	50.08	12	27.639
W2A1D1.jpg	47.83	12	28.064
W2A1D2.jpg	38.25	12	31.329
Total	45.39	36	28.691

Figure 6.10 - Results of the Kruskal-Wallis H Test for W2A1

The distributions of bias for each distortion were not similar for all groups. The values of bias reported for each distortion differed from one another, but the differences were not statistically significant, $H = X^2(2) = 1.224$, p = 0.542. The null hypothesis is retained.

6.3.9 W2A2 (D0/D1/D2) – The Guardian, Life Expectancy



bias			
Image Being Tested	Mean	Ν	Std. Deviation
W2A2D0.jpg	26.64	14	30.851
W2A2D1.jpg	26.64	14	28.830
W2A2D2.jpg	7.54	13	8.978
Total	20.59	41	26.166

Figure 6.11 - Results of the Kruskal-Wallis H Test for W2A2

The distributions of bias for each distortion were not similar for all groups. The values of bias reported for each distortion differed from one another, but the differences were not statistically significant, $H = X^2(2) = 5.638$, p = 0.060. The null hypothesis is retained.

6.3.10 Overview of Results

There was no statistically significant difference found between the levels of bias reported for articles presented with or without graphics. In all nine cases, the null hypothesis, that the bias for each distortion was not statistically different from the other distortions, was retained. These results are discussed more fully in Section 7.1.

6.4 Bias Levels by News Story

This section details the outcome of the Kruskal-Wallis H Test for each combination of website and distortion. This directly compares the bias reported for each story, on the same website and for the same level of distortion.



Bias Reported			
Image Being Tested	Mean	Ν	Std. Deviation
W0A0D0.jpg	38.92	13	30.404
W0A1D0.jpg	45.88	17	28.295
W0A2D0.jpg	16.50	12	10.775
Total	35.33	42	27.716

Figure 6.12 - Results of the Kruskal-Wallis H Test for W0D0

The distributions of bias for each distortion were not similar for all groups. The values of bias reported for each article differed significantly from one another, $H = X^2(2) = 7.393$, p = 0.025. The null hypothesis is <u>rejected</u>.

This indicates that a statistically significant difference has been detected between the levels of bias reported for the each of the Daily Mail articles. It can be seen that the article about life expectancy (A2) was considered the least biased. This result is particularly important when compared to the other results for other distortions of the Daily Mail.

6.4.2 W0D1 (A0/A1/A2) – The Daily Mail, original graphic



Bias Reported			
Image Being Tested	Mean	N	Std. Deviation
W0A0D1.jpg	53.07	14	34.582
W0A1D1.jpg	26.44	16	32.737
W0A2D1.jpg	27.67	12	31.938
Total	35.67	42	34.646

Figure 6.13 - Results of the Kruskal-Wallis H Test for W0D1

The distributions of bias for each distortion were not similar for all groups. The values of bias reported for each distortion differed from one another, but the differences were not statistically significant, $H = X^2(2) = 4.913$, p = 0.086. The null hypothesis is retained.



Bias Reported	
---------------	--

Image Being Tested	Mean	Ν	Std. Deviation
W0A0D2.jpg	47.00	14	22.185
W0A1D2.jpg	32.50	14	32.938
W0A2D2.jpg	36.17	12	38.364
Total	38.68	40	31.326

Figure 6.14 - Results of the Kruskal-Wallis H Test for WOD2

The distributions of bias for each distortion were not similar for all groups. The values of bias reported for each distortion differed from one another, but the differences were not statistically significant, $H = X^2(2) = 2.832$, p = 0.243. The null hypothesis is retained.

6.4.4 W1D0 (A0/A1/A2) – The BBC, no graphic



Bias Reported			
Image Being Tested	Mean	N	Std. Deviation
W1A0D0.jpg	43.58	12	38.242
W1A1D0.jpg	27.29	14	30.040
W1A2D0.jpg	36.86	14	29.294
Total	35.53	40	32.316

Figure 6.15 - Results of the Kruskal-Wallis H Test for W1D0

The distributions of bias for each distortion were not similar for all groups. The values of bias reported for each distortion differed from one another, but the differences were not statistically significant, $H = X^2(2) = 1.175$, p = 0.556. The null hypothesis is retained.



Bias Reported			
Image Being Tested	Mean	N	Std. Deviation
W1A0D1.jpg	28.08	12	25.040
W1A1D1.jpg	37.69	13	34.565
W1A2D1.jpg	49.12	17	29.801
Total	39.57	42	30.692

6.4.5 W1D1 (A0/A1/A2) – The BBC, original graphic

Figure 6.16 - Results of the Kruskal-Wallis H Test for W1D1

The distributions of bias for each distortion were not similar for all groups. The values of bias reported for each distortion differed from one another, but the differences were not statistically significant, $H = X^2(2) = 3.726$, p = 0.155. The null hypothesis is retained.

6.4.6 W1D2 (A0/A1/A2) – The BBC, experiment graphic



Bias Reported			
Image Being Tested	Mean	Ν	Std. Deviation
W1A0D2.jpg	36.17	12	25.665
W1A1D2.jpg	43.93	14	29.531
W1A2D2.jpg	37.69	16	33.738
Total	39.33	42	29.679

Figure 6.17 - Results of the Kruskal-Wallis H Test for W1D2

The distributions of bias for each distortion were not similar for all groups. The values of bias reported for each distortion differed from one another, but the differences were not statistically significant, $H = X^2(2) = 0.79$, p = 0.688. The null hypothesis is retained.

6.4.7 W2D0 (A0/A1/A2) – The Guardian, no graphic



Bias Reported			
Image Being Tested	Mean	Ν	Std. Deviation
W2A0D0.jpg	26.25	16	31.118
W2A1D0.jpg	50.08	12	27.639
W2A2D0.jpg	26.64	14	30.851
Total	33.19	42	31.272

Figure 6.18 - Results of the Kruskal-Wallis H Test for W2D0

The distributions of bias for each distortion were not similar for all groups. The values of bias reported for each distortion differed from one another, but the differences were not statistically significant, $H = X^2(2) = 5.343$, p = 0.069. The null hypothesis is retained.

6.4.8 W2D1 (A0/A1/A2) – The Guardian, original graphic



Bias Reported			
Image Being Tested	Mean	Ν	Std. Deviation
W2A0D1.jpg	26.21	14	28.631
W2A1D1.jpg	47.83	12	28.064
W2A2D1.jpg	26.64	14	28.830
Total	32.85	40	29.515

Figure 6.19 - Results of the Kruskal-Wallis H Test for W2D1

The distributions of bias for each distortion were not similar for all groups. The values of bias reported for each distortion differed from one another, but the differences were not statistically significant, $H = X^2(2) = 4.788$, p = 0.091. The null hypothesis is retained.

6.4.9 W2D2 (A0/A1/A2) – The Guardian, experiment graphic



Blas Reported			
Image Being Tested	Mean	N	Std. Deviation
W2A0D2.jpg	27.94	17	25.445
W2A1D2.jpg	38.25	12	31.329
W2A2D2.jpg	7.54	13	8.978
Total	24.57	42	26.287

Figure 6.20 - Results of the Kruskal-Wallis H Test for W2D2

. . . .

The distributions of bias for each distortion were not similar for all groups. However, the values of bias reported for each article differed significantly from one another, $H = X^2(2) = 12.378$, p = 0.002. The null hypothesis is <u>rejected</u>.

This indicates that a statistically significant difference was detected in the reported levels of bias for each article in The Guardian at the D2 level of distortion. The level of bias reported for the article about life expectancy (A2), was significantly lower than for the other two articles. This result is particularly important when compared to the results in Sections 6.4.7 and 6.4.8, where the differences were not statistically significant for the other levels of distortion of The Guardian website.

6.4.10 Overview of Results

There was a statistically significant difference detected between the bias reported for the Daily Mail at the D0 level of distortion, and for The Guardian at the D2 level of distortion. There was no significantly significant difference detected between the bias ratings of articles for any other website or level of distortion. Although this test was specifically a comparison of different articles for each website, at each level of distortion, a further inference can be made. Given that the only difference between the presentations within the group of sections 6.4.1 - 6.4.3, within the group of sections 6.4.4 - 6.4.6 and also within the group of sections 6.4.7 - 6.4.9, is the level of distortion, and the H-statistic for each of these tests is different (sometimes notably different), this tells us that the level of distortion (D0 / D1 / D2) is having an effect on the level of bias perceived. This indicates that the graphics accompanying the news articles have an effect on the level of bias perceived for the article as a whole.

6.5 Bias Levels by Website

This section details the outcome of the Kruskal-Wallis H Test for each combination of article and distortion. This directly compares the bias reported for each website's reporting of the same story, and at the same level of distortion.



Bias Reported			
Image Being Tested	Mean	N	Std. Deviation
W0A0D0.jpg	38.92	13	30.404
W1A0D0.jpg	43.58	12	38.242
W2A0D0.jpg	26.25	16	31.118
Total	35.34	41	33.170

6.5.1 A0D0 (W0/W1/W2) – CPI Inflation, no graphics

Figure 6.21 - Results of the Kruskal-Wallis H Test for A0D0

The distributions of bias for each distortion were not similar for all groups. The values of bias reported for each distortion differed from one another, but the differences were not statistically significant, $H = X^2(2) = 1.781$, p = 0.411. The null hypothesis is retained.

6.5.2 AOD1 (W0/W1/W2) – CPI Inflation, original graphics



Bias Reported			
Image Being Tested	Mean	N	Std. Deviation
W0A0D1.jpg	53.07	14	34.582
W1A0D1.jpg	28.08	12	25.040
W2A0D1.jpg	26.21	14	28.631
Total	36.18	40	31.733

Figure 6.22 - Results of the Kruskal-Wallis H Test for A0D1

The distributions of bias for each distortion were not similar for all groups. The values of bias reported for each distortion differed from one another, but the differences were not statistically significant, $H = X^2(2) = 4.722$, p = 0.094. The null hypothesis is retained.

6.5.3 A0D2 (W0/W1/W2) – CPI Inflation, experiment graphics



Blas Reported			
Image Being Tested	Mean	Ν	Std. Deviation
W0A0D2.jpg	47.00	14	22.185
W1A0D2.jpg	36.17	12	25.665
W2A0D2.jpg	27.94	17	25.445
Total	36.44	43	25.257

Figure 6.23 - Results of the Kruskal-Wallis H Test for A0D2

The distributions of bias for each distortion were not similar for all groups. The values of bias reported for each distortion differed from one another, but the differences were not statistically significant, $H = X^2(2) = 4.730$, p = 0.094. The null hypothesis is retained.

6.5.4 A1D0 (W0/W1/W2) – Bitcoin Prices, no graphics



Bias Reported			
Image Being Tested	Mean	Ν	Std. Deviation
W0A1D0.jpg	45.88	17	28.295
W1A1D0.jpg	27.29	14	30.040
W2A1D0.jpg	50.08	12	27.639
Total	41.00	43	29.670

Figure 6.24 - Results of the Kruskal-Wallis H Test for A1D0

The distributions of bias for each distortion were not similar for all groups. The values of bias reported for each distortion differed from one another, but the differences were not statistically significant, $H = X^2(2) = 4.763$, p = 0.092. The null hypothesis is retained.

6.5.5 A1D1 (W0/W1/W2) – Bitcoin Prices, original graphics



Dias Reported			
Image Being Tested	Mean	Ν	Std. Deviation
W0A1D1.jpg	26.44	16	32.737
W1A1D1.jpg	37.69	13	34.565
W2A1D1.jpg	47.83	12	28.064
Total	36.27	41	32.502

Figure 6.25 - Results of the Kruskal-Wallis H Test for A1D1

Dias Donartad

The distributions of bias for each distortion were not similar for all groups. The values of bias reported for each distortion differed from one another, but the differences were not statistically significant, $H = X^2(2) = 3.635$, p = 0.162. The null hypothesis is retained.

6.5.6 A1D2 (W0/W1/W2) – Bitcoin Prices, experiment graphics



Bias Reported			
Image Being Tested	Mean	N	Std. Deviation
W0A1D2.jpg	32.50	14	32.938
W1A1D2.jpg	43.93	14	29.531
W2A1D2.jpg	38.25	12	31.329
Total	38.23	40	30.864

Figure 6.26 - Results of the Kruskal-Wallis H Test for A1D2

The distributions of bias for each distortion were not similar for all groups. The values of bias reported for each distortion differed from one another, but the differences were not statistically significant, $H = X^2(2) = 1.365$, p = 0.505. The null hypothesis is retained.

6.5.7 A2D0 (W0/W1/W2) – Life Expectancy, no graphics



Bias Reported			
Image Being Tested	Mean	N	Std. Deviation
W0A2D0.jpg	16.50	12	10.775
W1A2D0.jpg	36.86	14	29.294
W2A2D0.jpg	26.64	14	30.851
Total	27.17	40	26.549

Figure 6.27 - Results of the Kruskal-Wallis H Test for A2D0

The distributions of bias for each distortion were not similar for all groups. The values of bias reported for each distortion differed from one another, but the differences were not statistically significant, $H = X^2(2) = 3.730$, p = 0.155. The null hypothesis is retained.

6.5.8 A2D1 (W0/W1/W2) – Life Expectancy, original graphics



Bias Reported			
Image Being Tested	Mean	Ν	Std. Deviation
W0A2D1.jpg	27.67	12	31.938
W1A2D1.jpg	49.12	17	29.801
W2A2D1.jpg	26.64	14	28.830
Total	35.81	43	31.327

Figure 6.28 - Results of the Kruskal-Wallis H Test for A2D1

The distributions of bias for each distortion were not similar for all groups. However, the values of bias reported for each article differed significantly from one another, $H = X^2(2) = 7.847$, p = 0.020. The null hypothesis is <u>rejected</u>.

This means that a statistically significant difference was detected between the levels of bias reported for the different website's reporting on life expectancy, with the articles presented in their original format with the original graphic. The BBC had a significantly higher level of perceived bias than the Guardian in their article. This result is particularly important when taken in comparison to the results detailed in Sections 6.5.7 and 6.5.9. For those other results, no difference of statistical significance was detected, suggesting that the level of distortion had an effect on the level of bias perceived for the article.



W1A2D2.jpg



Image Being Tested

W2A2D2.jpg

Figure 6.29 - Results of the Kruskal-Wallis H Test for A2D2

The distributions of bias for each distortion were not similar for all groups. The values of bias reported for each distortion differed from one another, but the differences were not statistically significant, $H = X^2(2) = 5.191$, p = 0.075. The null hypothesis is retained.

6.5.10 Overview of Results

60.00

40.00

20.00-

0.00-

W0A2D2.ipc

There was a statistically significant difference detected between the three websites' reporting the A2 article, at the D1 level of distortion. No other combinations of article and distortion produced any significantly different results between websites.

Although this test was specifically a comparison of different websites for each article, at each level of distortion, a further inference can be made. Given that the only difference between the presentations within the group of sections 6.5.1 – 6.5.3, within the group of sections 6.5.4 - 6.5.6 and also within the group of sections 6.5.7 - 6.5.9, is the level of distortion, and the Hstatistic for each of these tests is different (sometimes notably different), this tells us that the level of distortion (D0 / D1 / D2) is having an effect on the level of bias perceived. This indicates that the graphics accompanying the news articles have an effect on the level of bias perceived for the article as a whole.

6.6 Summary of Results

Independent	Sample	H Test Result	<i>p</i> -value	Null Hypothesis
Variable				(REJECT if <i>p</i> -value
Being Tested				< 0.05)
D0, D1, D2	W0A0	1.451	0.484	Retain
	W0A1	4.832	0.112	Retain
	W0A2	0.164	0.921	Retain
	W1A0	1.180	0.554	Retain
	W1A1	2.593	0.274	Retain
	W1A2	1.983	0.371	Retain
	W2A0	1.036	0.596	Retain
	W2A1	1.224	0.542	Retain
	W2A2	5.638	0.060	Retain
A0, A1, A2	W0D0	7.393	0.025	REJECT
	W0D1	4.913	0.086	Retain
	W0D2	2.832	0.243	Retain
	W1D0	1.175	0.556	Retain
	W1D1	3.726	0.155	Retain
	W1D2	0.79	0.688	Retain
	W2D0	5.343	0.069	Retain
	W2D1	4.788	0.091	Retain
	W2D2	12.378	0.002	REJECT
W0, W1, W2	A0D0	1.781	0.411	Retain
	A0D1	4.722	0.094	Retain
	A0D2	4.370	0.094	Retain
	A1D0	4.763	0.092	Retain
	A1D1	3.635	0.162	Retain
	A1D2	1.365	0.505	Retain
	A2D0	3.370	0.155	Retain
	A2D1	7.847	0.020	REJECT
	A2D2	5.191	0.075	Retain

Table 6.3 – A summary of the key statistics generated by the results for this experiment, and the decision to retain or reject the null hypothesis

6.7 Conclusions

6.7.1 Bias Rating Results

Although differences were found in the levels of bias reported between different groups, for only three of the twenty-seven permutations of this experiment was that difference considered statistically significant. There is no obvious trend in the rejection of the null hypothesis for these samples.

Crucially, for answering the research question, none of the differences found in relation to the different levels of graphical distortion (D0, D1, D2) was found to be statistically significant.

Therefore, no statistically significant evidence has been found to support the proposition that the level of perceived bias can be affected by the graphics accompanying a news article.

6.7.2 Level of Significance

While the differences were determined not to be statistically significant, there were small differences between each group tested. There were some which were close to achieving statistical significance; eleven of the twenty-seven tests had *p*-values of <0.1, as seen in Figure 6.30.



Figure 6.30 – The p-values for each of the tests carried out on the bias ratings. The horizontal red line signifies p=0.1

The limit of p=0.05, which was used in this experiment as the determining factor for significance, is the standard used for such tests. However, this value is somewhat arbitrary, and while rejection of the null hypothesis would have given a clear result, the retention of the null hypothesis in most of these cases does not tell the whole story. Differences do exist

between the different groups, and these are discussed in Chapter 7. Also, a wealth of other data was also collected during the experiment on the participants' backgrounds, which is also discussed in Chapter 7.

7. Discussion of Results

This chapter will focus on analysis of the results detailed in Chapter 6, and will also contain discussion about the other data obtained in the survey section of the experiment.

The full set of responses recorded from the participants is given in Appendix A1 and Appendix A2.

7.1 Bias Rating Results

The results obtained from this experiment, as detailed in Chapter 6, show that graphics accompanying news articles can have some impact on the level of bias perceived in the articles. However, the tests in this experiment found many of these effects to be very small, and very little evidence of these effects being statistically significant.

The tests which directly examine the difference between bias levels for different distortions of the same website and articles found no statistically significant difference across any of the distortion levels. There were differences observed in seven of the nine such tests, but these were too small to be deemed statistically significant.

7.1.1 The Daily Mail

For each article taken from the Daily Mail (W0), the addition of a graphic had a different effect, one increasing the perceived bias (A0), one reducing the perceived bias (A1), and one making no discernible difference (A2), as can be seen in Figures 6.1 - 6.3. In both cases where the addition of a graphic caused a large change in the level of bias, the second distortion (D2) actually served to moderate this effect; the second distortion was never considered the most or least biased, but was actually roughly half way between. Although the different distortions did not much affect the mean-rank for the third article (A2), there were very different distributions for the results of each, with the responses becoming more distributed with each distortion.

It is clear, therefore that while the graphic had some effect on the perceived bias of the articles, what that effect was depended on the articles and graphics.

7.1.2 The BBC

Similar results were seen across the samples taken from the BBC (W1), as seen in Figures 6.4 - 6.6. Here, again, the addition of a graphic changed the bias level where compared to when the article was presented without a graphic. Once again, the second level of distortion was never perceived as being the most biased. Also, similar to the results for the Daily Mail was

the result that for the third article (A2), the most significant change in bias levels was that the responses became more widely distributed as the distortion level increased.

Again, it is clear that the graphic accompanying the articles had an effect on the perceived level of bias for that article. It is notable that, like for the Daily Mail, the graphic for the D2 level of distortion served to divide opinion among participants rather than to change the overall average of their responses.

7.1.3 The Guardian

Finally, for the Guardian (W2), bias levels appear to have been affected to a much smaller extent by the distortions, as seen in Figure 6.7 - 6.9. Overall, the responses to these samples had a much smaller distribution, indicating larger consensus among participants on the presence of bias in the samples.

It is particularly notable that the difference between the D0 level of distortion, with no graphic, and the D1 level of distortion, with original graphic, was negligible across all three articles. This indicates that just the addition of the graphic alone did not serve to alter the level of bias perceived to be present in the article it accompanied.

7.1.4 Website and Article Dimensions

Tests across the other dimensions of the experiment cube further underline that the graphics accompanying news articles do have some effect on the perceived bias of the article, as the relative differences between articles and websites changes, sometimes significantly, between tests on different levels of distortion.

For example, as seen in Figure 6.10, there is a statistically significant difference between the levels of bias reported for each article. As seen in Figure 6.11 and Figure 6.12, for the other levels of distortion, the difference between the groups is not statistically significant. This is possible only by the relative change in bias owing to the changes in the distortion, as the articles and websites remain the same.

Similar conclusions can be drawn for both other cases where the null hypothesis was rejected (W2D2 and (A2D1).

7.1.5 Impact of Distortions

The differences observed between the distortions with no graphic present (D0), and the distortions with graphics added (D1 and D2), were not statistically significant, but still warrant explanation.

The difference is unlikely to be simply a matter of the website design changing, as it can be seen in Figures 6.3, 6.7, 6.8, and 6.9, that there is no considerable change in the bias rating

with the addition of a graphic (D1), compared to when there was none (D0). In three of these cases (Figure 6.3, 6.8, and 6.9), however, there is a discernible change in the level of perceived bias between the D1 and D2 levels of distortion. This could be an indication that the level of perceived bias has been affected by the graphic itself, and as such that information bias within the graphic itself has affected the overall perceived level of bias for the article.

7.1.6 Impact of Graphics

The differences evident in the box-plots of results for the Kruskel-Wallis H Test, shown in Figure 6.3 - 6.29, show that the perceived level of bias does change depending on the graphic present. The failure to reject the null-hypothesis in most cases in this experiment may be put down to the sample size being too small to detect this effect size; as the sample size increases, so too does the proximity of a *p*-value to its true value. It is fully possible that the differences observed in this experiment were too small to detect statistically with the sample sizes used. This does not make the evidence of differences irrelevant; they just can't be deemed statistically significant at a 95% confidence level for this particular sample size.

7.2 Hypothesis Results

The null hypothesis for this experiment stated:

(H_0) This research finds no evidence of bias being introduced to a news article through the use of an information graphic.

The alternative hypothesis stated:

(H_{A}) This research finds evidence of bias being introduced to a news article through the use of an information graphic.

This experiment found no statistically significant evidence to support the proposition that information bias in the graphics accompanying news articles online can affect the level of perceived bias for the article as a whole. Therefore, the null hypothesis is retained.

7.3 Bias Attributed to Publication

It was clear at the design stage of the experiment, that recognition, and opinions, among participants of the websites used in the experiment, could affect the participants' judgement of a website. It was for this reason that participants were asked to report any websites they recognised, and any websites they considered particularly biased. This section analyses those responses.

Every effort was made in the design of this experiment to minimise the effects of participants having pre-conceived notions of how biased they expect a particular news website to be, and

rating the bias accordingly. This is a known effect, as discussed in Chapter 2, and was to be avoided, insofar as possible, in order to obtain accurate ratings for the specific samples used in this experiment. Therefore, every precaution was taken to avoid it.

For this experiment all of the news websites were from the UK, and all of the participants were US citizens living in the USA. Additionally, all names and branding were removed from the images of websites shown to the participants. They could not determine the website without prior knowledge of it. That said, owing to the universal nature of publications on the internet, it is impossible to categorically find a demographic who will not know a certain news agency.

7.3.1 Participant Recognition of Websites

Recognising this fact, and in an attempt to quantify any effect arising from it, at the end of the experiment, participants were asked to select any publications they had recognised during the experiment, from a multiple-choice list. The numbers recognising publications is shown in Table 7.1.

There were three websites used in the experiment, but nine on the multiple-choice list for participants. This was to ensure that participants truly did recognise the websites used and could differentiate them from websites not used. By only having the three used in the experiment on the list, participants could have convinced themselves that they had recognised each, in a sort of confirmation bias of itself. By having to differentiate between websites, a participant is required to have genuinely recognised the website, or thought they had.

Website	Number of Participants	Percentage of Participants
None	54	43.55%
The Guardian*	43	34.68%
The Telegraph	25	20.16%
The Independent	22	17.74%
The Daily Mail*	48	38.71%
The Spectator	7	5.56%
New Statesman	6	4.84%
Al Jazeera	23	18.55%
BBC*	47	37.90%
Reuters	25	20.16%

Table 7.1 – News-websites reportedly recognised by participants. The actual websites used are highlighted in grey and with an asterisk.

It is evident, from the data collected, that a small number of these responses could be disregarded. Specifically, five respondents claimed to have recognised all nine websites

during the course of the experiment, which would have been impossible as each participant was only presented with six (two instruction tasks, one spike question, and three in the experiment proper). In these cases, it is likely that they just selected all of the boxes on the multiple-choice survey question without regard to the question being asked or the responses being selected.

There are many cases evident in Table 7.1 where participants incorrectly identified websites which were not a part of the experiment. This could be attributed to mistakenly misattributing certain design features from a website not in the experiment, to one on the list. It could also simply be a case of mistaking a websites name for another.

Even taking this into account, it is noticeable that the most recognised websites by far were in fact the ones used in the experiment, as seen in Figure 7.1. These are The Guardian (W2), The Daily Mail (W0) and the BBC (W1), with well over one-third of participants recognising each of those websites.



Figure 7.1 – Number of participants claiming to recognise each website during the experiment, chosen from a multiple-choice list.

While the greatest number of responses, almost 44%, claimed to recognise none of the websites, the majority claimed to have recognised one or more. This shows that the efforts made to reduce this effect were not entirely successful.

Furthermore, it can be seen in Figure 7.2 that over one-fifth, of the 124 respondents, correctly identified all three websites.



Figure 7.2 – Number of participants correctly identifying a number of websites.

With this level of recognition for the websites used, quantification of any effect this may have had on the experiment was important. This was made possible by using the responses to another of the reflective questions posed to participants.

7.3.2 Participants' Opinions of the Websites' Bias

Participants were also asked if they considered any of the news websites on the list to be particularly biased in general. The results can be seen in Table 7.2.

Website	Number of Participants	Percentage of Participants
None	61	49.19%
The Guardian *	20	16.13%
The Telegraph	14	11.29%
The Independent	9	7.26%
The Daily Mail *	51	41.13%
The Spectator	7	5.56%
New Statesman	6	4.84%
Al Jazeera	18	14.52%
BBC *	12	9.68%
Reuters	8	6.45%

Table 7.2 – The number of participants who consider each news website to be particularly biased.The websites used in this experiment are highlighted in grey and with an asterisk.

Particularly of note from these results is that over two-fifths of participants considered the Daily Mail to be particularly biased. As this is one of the news websites used in this experiment, and almost two-fifths claimed to recognise it when they saw it in the experiment,

it raises the possibility that participants' opinions of the websites, external to this experiment, may have influenced the results. Participants could only respond in this way if they had some prior knowledge or opinions on the website.



Figure 7.3 – The number of participants who consider each news website to be particularly biased.

7.3.3 Effect on Bias Ratings

In order to investigate if this may have been the case, further analysis was required. For this, participants were split into three separate groups for each website:

- 1. Those who did not recognise the website being tested.
- 2. Those who recognised the website being tested, but do not consider that website to be particularly biased.
- 3. Those who recognised the website being tested, and who consider that website to be particularly biased.

The bias rating that each participant gave for each website, regardless of distortion or story, was considered. To determine if there was an effect from websites being recognised, the relative levels of bias reported was compared across the three groups for each website, and a determination was made as to the significance of this.

As with the bias results for the main part of the experiment, the data involved does not satisfy the assumptions required for ANOVA, as significant outliers exist in the bias ratings. The data does satisfy the assumptions for the Kruskel-Wallis H Test, and so this test was used once again.

The null hypothesis for these tests assumed that the distributions for each group would be equal, indicating that recognition and prior expectation of the website had no impact on the bias reported by the participant.

Rejection of the null hypothesis would indicate that recognition of the websites, and an expectation of that website's bias, had influenced the level of bias reported by the participants.

7.3.4 Results of Bias Testing

The groups were analysed for each website, and the results are as follows.

7.3.4.1 Daily Mail (W0)



Bias Reported			
Group	Mean	N	Std. Deviation
Website Not Recognised	39.01	78	31.121
Website Recognised, Not Considered Biased	13.31	16	13.817
Website Recognised, Considered Biased	41.40	30	32.405
Total	36.27	124	30.934

Figure 7.4 - Results of the Kruskal-Wallis H Test for the Daily Mail (WO)

The distributions of bias for each group were not similar for all groups. The values of bias reported by each group differed significantly from one another, $X^2(2) = 10.373$, p = 0.006. The null hypothesis is **rejected**, indicating that existing opinions of the website's bias significantly affected the results of this experiment.



Bias Reported			
Group	Mean	N	Std. Deviation
Website Not Recognised	36.10	77	30.144
Website Recognised, Not Considered Biased	36.97	39	29.534
Website Recognised, Considered Biased	56.75	8	32.766
Total	37.71	124	30.289

Figure 7.5 - Results of the Kruskal-Wallis H Test for the BBC (W1)

The distributions of bias for each group were not similar for all groups. The values of bias reported by each group differed, but not significantly, $X^2(2) = 3.290$, p = 0.193. The null hypothesis is retained, indicating that existing opinions of the website's bias did not significantly affect the results of this experiment.
7.3.4.2 The Guardian (W2)



Bias Reported			
Group	Mean	N	Std. Deviation
Website Not Recognised	31.29	80	28.798
Website Recognised, Not Considered Biased	20.65	31	26.649
Website Recognised, Considered Biased	44.23	13	33.917
Total	29.98	124	29.371

Figure 7.6 - Results of the Kruskal-Wallis H Test for The Guardian (W2)

The distributions of bias for each distortion were not similar for all groups. The values of bias reported by each group differed significantly from one another, $X^2(2) = 6.202$, p = 0.045. The null hypothesis is <u>rejected</u>, indicating that existing opinions of the website's bias significantly affected the results of this experiment.

7.3.5 Analysis of Results for bias attributed to publications

It can be seen from these results that participants' existing opinions pertaining to the perceived bias of each website has had a statistically significant impact on the results. While there is no significant impact of this evident for responses to the BBC's articles, it is worth noting that the number of participants who considered that website to be biased was relatively low anyway, thus increasing the *p*-value and making rejection of the null hypothesis less likely, regardless of the responses.

Similarly, the difference between groups is most significant for the Daily Mail, which was considered biased by the largest number of participants of any website. While this increases the significance in itself, the simple fact that knowledge of the website has had an effect on results, and that approximately two-fifths of respondents recognised it and considered it biased, shows that this did in some way affect the results obtained for bias ratings.

This additional bias, introduced by the participant, is a confounding variable. The effect was known and acknowledged by the experiment design; it was for this reason that participants were asked the reflective questions. This does pose a challenge to carrying out experiments of this type, and perhaps other experiment designs could be considered to reduce the effect further. For example, participants could be asked to identify websites they would recognise before they start rating them, and then only be presented with those websites which they would not recognise.

The results also provide further evidence that consumers tend to favour news media which they find agreeable. By far the lowest levels of bias were reported for websites which were recognised by participants, but which those participants did not consider to be biased.

7.4 Other External Factors

Consideration was also given to other information supplied by the participants, and how that might have affected their responses. No correlation was found between the level of bias reported and the age, level of education, occupation, or expected yearly salary reported by the participants. These social and economic factors therefore likely played no role in altering the levels of bias reported by the participants. Visualisation of these results are shown in Appendix A6.

There is no apparent correlation between political leaning and the level of bias reported by the participants. With almost three-quarters of participants reporting that they hold a centre or left-of-centre political view, any effect which is caused by political leaning will be somewhat reduced by the homogeneity of the participant's viewpoints anyway.

There is also no apparent correlation between the amount of time participants spend online each day or the amount of time the participants spend reading the news online each day and the level of bias reported by participants in the experiment. This indicates that familiarity with accessing news online played no significant part in altering participants' perception of bias.

The vast majority of participants reported, in a multiple-choice survey question, regularly consuming news online, either using websites or social media the results of which are shown in Figure 7.7.



Figure 7.7 – Media regularly used by participants to consume news.

7.5 Factors Contributing to Bias

The final question asked of participants in this experiment was, "When judging bias in news articles online, which of the following do you consider to be significant elements of bias?"

This question was designed to elicit what they believe constitutes bias in news media. There is very little information in existing literature which ascertains exactly how people understand bias. This likely contributes to confusion in the domain.

It should be noted that the participants had already been presented with the definition of bias for this experiment before answering this question, that is:

"Deliberate or accidental slant by the journalist, editor or publication to distort reality".

7.5.1 Responses

Figure 7.8 shows the responses received to this multiple-choice question.



Figure 7.8 – Responses to the question, "When judging bias in news articles online, which of the following do you consider to be significant elements of bias?"

The letters A-I in the bar chart in Figure 7.8 correspond to the following options:

- A = "Lack of balance"
- B = "Unfairness"
- C = "Subjectivity"
- D = "Distortion affecting one side of a story"
- E = "Not giving one side of a story as much coverage as the other"
- F = "Expression of personal opinions by a journalist"
- G = "Providing sources for one side of a story and not the other"
- H = "Providing data to support one side of a story and not the other"
- I = "Order in which sides of a story are reported"

All of the possible responses to this survey question were intentionally unqualified, requiring a level of interpretation on the part of the participant. The responses provide insight to what the participants in this experiment were judging their perception of bias.

7.5.2 Analysis of Responses

Of particular note are the number of participants who chose options G and H. Over 22% more participants chose option H, when compared to option G. This suggests that they consider it more important to provide data for both sides of a story than to provide sources. This perhaps highlights a distrust in the media generally, where readers would apparently rather make sense of the raw information themselves, than to have it interpreted by a journalist or other person who is cited as being the source.

Also interesting is that option B, "Unfairness", was selected by far fewer participants than most of the other responses. Just over one-third of the 124 of participants chose this option compared to over one-half selecting many of the other options. "Unfairness" is an English-language synonym for "bias", making the lack of participants choosing it somewhat peculiar. Many of the options which were most-selected were more specifically tailored to news media, and being less vague may have seemed more relevant to the question.

Most specifically, it is interesting to note how significantly participants' opinions of what constituted bias differed from participant to participant, despite each participant taking the same survey during which they were each presented with the same definition of bias. This highlights a challenge in accurately carrying out this sort of experiment, where participant subjectivity, and participant interpretation of the questions, can play a large role in forming the participants' answers. Even by presenting participants with a common definition for bias, their subjectivity can play a part in how they apply that definition.

7.6 Summary of Results

The null hypothesis for this experiment was retained, as there was no statistically significant evidence to support the proposition that information bias in the graphics accompanying news articles affected the level of bias perceived for the article as a whole. Despite this, there was a small effect evident in the results, which was too small to confirm statistically in this experiment.

More importantly, it was found that, despite efforts made to minimise the chances of this happening, participants' pre-determined opinions of the news websites presented in the experiment significantly affected how they responded to the survey. Those who expected a website to be particularly biased reported a significantly higher bias rating for that website than other participants for the same website. This undermines the confidence we can have that the results to the bias ratings are a fair and considered response to what was presented

to participants, and that the research into the different levels of graphical distortion has been compromised by bias on the part of a significant number of the participants.

No other external factors seem to have influenced the results to any considerable degree, and participants were almost all quite familiar with the online news medium.

The final finding shows a lack of consensus among participants of what constituted bias. This is in spite of the participants all working from the same definition of bias, and suggests that participants reported bias ratings for the articles based on different criteria.

8. Conclusions

This research builds on a significant body of existing research into media bias, asking specifically the question:

"To what extent do the information graphics accompanying a news article impact upon the perceived bias of the article?"

The research indicates that the information graphics accompanying a news article can have a material impact on the perceived bias of that article. The effect size observed in this particular experiment was deemed statistically insignificant however. There was also no clear trend in the results of this experiment, in terms of whether the bias level increased or decreased for different levels of distortion. Despite this, visualisations of the data show that graphics do impact on the perceived bias, suggesting that the research topic may be worthy of further investigation.

A more significant finding arising from the user survey carried out, was that despite efforts to minimise the chances of participants recognising the news websites used in the experiment, and bringing their own opinions of those websites to bear on the results of the experiment, there is significant evidence to suggest that this was the case. This is a result of crowdsourcing participants, where the researchers have no control over what information participants do or do not have, external to that which is presented to them in the experiment. This is unavoidable, but must be noted.

Importantly, where participants recognised a website which they considered to be particularly biased, they affirmed their own opinion by rating the bias level for that website significantly higher than others who had no such opinion. Similarly, where participants recognised a website but did not expect it to be biased, the participants reported significantly lower levels of bias than those who had no such opinion. This is strong evidence of confirmation bias, where the participant actively looks for reasons to give a high bias rating to a website they consider to be biased. There are two important considerations resulting from this:

- 1. The results agree with existing research in the area, further adding to the evidence of the effects of confirmation bias.
- 2. Despite intentional efforts to minimise the effects, the confounding variable of participants having an opinion of the bias associated with certain websites, played a part in results. This presents a learning opportunity; a different experiment design should be considered for future similar work.

It was also interesting to find the lack of consensus among participants on what constitutes bias in news media. This lack of consensus came despite all participants having been presented with the same definition of bias for the experiment. Their responses serve to illustrate the multitude of factors which may affect a participant's determination on the presence of bias, and the difficulties which this presents to researchers in the area. This is an area which will require significant future work in order for a comprehensive understanding to be attained.

Although the null hypothesis was retained for this experiment, some evidence was found of an effect graphics had on the level of bias perceived for the article they accompany. Future work would be required to detect the precise significance of that effect.

8.1 Future Work and Refinements

The results of this experiment provide some interesting insights into the perceptions of bias in news media. Arising from this research are learning opportunities, and also areas which could benefit from further study. The lessons learned from this research should be taken into account in the formulation of the design of future experiments in order to avoid some of the confounding effects found in this research.

Every effort was made in the preparation of news website samples, to be rated by participants in the experiment, to ensure that the design of the website which was presented to participants was not significantly altered from its original design. While branding was removed, the rest of the website design and layout was left unaltered, in an attempt to avoid the known effects of website design on the perception of bias. Efforts were also made to minimise the possibilities of participants recognising the websites by having the source material and participants coming from different countries. In spite of this, a significant proportion of participants recognised the websites used, and with their own preconception of whether or not that website should be bias, skewed the results significantly in favour of their own opinions. This is not helpful to the research, as it does not compare like with like.

For future studies in this area, it should be considered, therefore, to present the articles alone without the rest of their website, or else in a common template which is not website specific. Another alternative, which would likely remove the effect altogether without removing design effects, would be to ask participants which websites they would recognise in advance of rating the images, and presenting them only with websites which they would not recognise. Whatever method is employed, it is clear that there is a need to remove the introduction of bias by the participants in this way.

Another possible refinement to the experiment relates to the samples used. The effect of the graphics on the articles tended to differ depending on the particular article and graphic. Presenting multiple graphics with the same article, or multiple articles on the same topic with and without the same graphic may provide further insight into what effect the graphics have. Varying the graphic, article and website may have been useful for providing an answer if the effect had been very large in this experiment, but as it was not, a more nuanced approach may be required which focusses on one specific aspect of this research.

In terms of effect size, it was seen in this research that the effect size was relatively small. In order to make a statistically significant finding for this using *p*-values, it would be necessary to use a much larger sample size. As the size increases the increasing accuracy of the *p*-value slows, meaning that it would possibly have to be much larger sample size than was used in this experiment.

Finally, further research could also be carried out into how users perceive bias based on what they consider to constitute bias. The findings of this research, as shown in Figure 7.8, suggest that there is far from consensus on the matter. Asking for bias ratings based on specific aspects of bias, rather than just from a general high-level definition, and examining the relative effect of these aspects could add significant understanding to how we determine that bias exists in a piece of news media.

8.2 Final Thoughts

Although the null hypothesis for this experiment was not rejected, the research has highlighted some important factors which can affect this sort of experiment which will be useful for future work. Effects including the introduction of bias through participants' existing experience of certain newspapers, despite significant efforts made to minimise this effect in the experiment design, serve to highlight the importance of ongoing consideration of experiment design for similar experiments. This was only detectable by collection of the appropriate data from participants, and regardless of the experiment design, these results have shown the importance of collecting this sort of background information to contextualise participants' responses.

While there was no statistically significant evidence found to support the supposition that information bias in the graphics accompanying news articles can affect the level of bias perceived for that article, finding exactly the extent to which the external factors influenced the results is an important finding in itself. Determining also that the effect of graphics on the bias in articles they accompany is likely quite small should provide important information for the design of future experiments researching this topic.

From these results it is evident that there is a strong likelihood that the graphics accompanying news articles affect the perception of bias in that article as a whole, but that this effect is relatively small, and requires further research to adequately confirm. This further research should explore the effects of different graphics, and the coverage of different topics, including those which may be emotive. Future work is also required to establish the effects of using different types of graphics.

References

- [1] M. Higgins, 'Mediated populism, culture and media form', *Palgrave Communications*, vol. 3, no. 1, p. 3, Oct. 2017.
- [2] E. C. Nisbet, P. Beck, and R. Gunther, 'Trump may owe his 2016 victory to "fake news," new study suggests', *The Conversation*. [Online]. Available: http://theconversation.com/trumpmay-owe-his-2016-victory-to-fake-news-new-study-suggests-91538. [Accessed: 10-Apr-2018].
- [3] 'Ethics in the News Media Lies and Brexit Ethical Journalism and the EU', *Ethical Journalism Network*.
- [4] T.-T. Lee, 'Why They Don't Trust the Media: An Examination of Factors Predicting Trust', *American Behavioral Scientist*, vol. 54, no. 1, pp. 8–21, Sep. 2010.
- [5] D. S. Tull, 'Intentional Bias in Public Opinion Polls for Decisional Purposes', *Public Opinion Quarterly*, vol. 39, no. 4, p. 552, 1975.
- [6] F. D. Russo, 'A Study of Bias in Tv Coverage of the Vietnam War: 1969 and 1970', *Public Opinion Quarterly*, vol. 35, no. 4, pp. 539–543, 1971.
- [7] T. Munzner, *Visualization analysis and design*, 1st ed. Boca Raton: CRC Press, 2014.
- [8] M. Tremayne, A. S. Weiss, and R. C. Alves, 'From product to service: the diffusion of dynamic content in online newspapers.', *Journalism & Mass Communication Quarterly*, vol. 84, no. 4, pp. 825–839, 2007.
- [9] J. Bar-Ilan, K. Keenoy, M. Levene, and E. Yaari, 'Presentation bias is significant in determining user preference for search results—A user study', J. Am. Soc. Inf. Sci., vol. 60, no. 1, pp. 135– 149, Jan. 2009.
- [10] J. A. Simpson and E. S. C. Weiner, Eds., *The Oxford English dictionary. Vol.2: B.B.C.-Chalypsography*, 2nd ed. Oxford: Clarendon P, 1989.
- [11] P. Waldman and J. Devitt, 'Newspaper photographs and the 1996 presidential election: The question of bias', *Journalism & Mass Communication Quarterly*, vol. 75, no. 2, pp. 302–311, 1998.
- [12] M. Gentzkow and J. M. Shapiro, 'Media Bias and Reputation', *journal of political economy*, p. 37.
- [13] Paul Andon and Clinton Free, 'Media coverage of accounting: the NRL salary cap crisis', *Acc Auditing Accountability J*, vol. 27, no. 1, pp. 15–47, Jan. 2014.
- [14] W. L. Bennett, *News: the politics of illusion*, 7th ed. Harlow: Longman, 2007.
- [15] J.-M. Eberl, H. G. Boomgaarden, and M. Wagner, 'One Bias Fits All? Three Types of Media Bias and Their Effects on Party Preferences', *Communication Research*, vol. 44, no. 8, p. 1125, Dec. 2017.
- [16] M. Kim, 'The Role of Partisan Sources and Audiences' Involvement in Bias Perceptions of Controversial News', *MEDIA PSYCHOLOGY*, vol. 19, no. 2, pp. 203–223, Apr. 2016.
- [17] T. Groeling, 'Media Bias by the Numbers: Challenges and Opportunities in the Empirical Study of Partisan News', *Annual Review of Political Science*, vol. 16, no. 1, pp. 129–151, May 2013.
- [18] T. E. Patterson and W. Donsbagh, 'News decisions: Journalists as partisan actors', *Political Communication*, vol. 13, no. 4, pp. 455–468, Oct. 1996.
- [19] E. Elejalde, L. Ferres, and E. Herder, 'On the nature of real and perceived bias in the mainstream media', *PLoS ONE*, vol. 13, no. 3, pp. 1–28, Mar. 2018.
- [20] A. W. Barrett and L. W. Barrington, 'Bias in newspaper photograph selection', *Political Research Quarterly*, vol. 58, no. 4, pp. 609–618, 2005.
- [21] R. Krestel, A. Wall, and W. Nejdl, 'Treehugger or petrolhead?: identifying bias by comparing online news articles with political speeches', in *Proceedings of the 21st international conference companion on World Wide Web*, New York, NY, USA, 2012, pp. 547–548.
- [22] M. Higgins, 'Mediated populism, culture and media form', *Palgrave Communications*, vol. 3, no. 1, p. 3, Oct. 2017.

- [23] E. C. . Nisbet nisbet. 5@osu. ed., M. Saldaña magdalenasaldan@gmail. co., T. J. . Johnson tom. johnson@austin. utexas. ed., G. Golan gjgolan@usf. ed., and A. Day daydoc@gmail. co., 'Credibility Gaps and Public Opinion in a Competitive Media Environment: The Case of Arab Satellite TV News in Lebanon', *International Journal of Communication (19328036)*, vol. 11, pp. 3072–3095, Jan. 2017.
- [24] T. GROELING, 'Who's the Fairest of them All? An Empirical Test for Partisan Bias on ABC, CBS, NBC, and Fox News', 2008.
- [25] A. Westerwick, B. K. Johnson, and S. Knobloch-Westerwick, 'Confirmation biases in selective exposure to political online information: Source bias vs. content bias', *Communication Monographs*, vol. 84, no. 3, pp. 343–364, Jul. 2017.
- [26] G. W. Yun, S.-Y. Park, S. Lee, and M. A. Flynn, 'Hostile Media or Hostile Source? Bias Perception of Shared News', *SOCIAL SCIENCE COMPUTER REVIEW*, vol. 36, no. 1, pp. 21–35, Feb. 2018.
- [27] Q. Ma and M. Yoshikawa, 'Topic and Viewpoint Extraction for Diversity and Bias Analysis of News Contents', in Advances in Data and Web Management, Q. Li, L. Feng, J. Pei, S. X. Wang, X. Zhou, and Q.-M. Zhu, Eds. Springer Berlin Heidelberg, 2009, pp. 150–161.
- [28] S. Park, M. Ko, J. Kim, H. Choi, and J. Song, 'NewsCube2. 0: An Exploratory Design of a Social News Website for Media Bias Mitigation', in *Workshop on Social Recommender Systems*, 2011.
- [29] R. M. Entman, 'Framing: Toward Clarification of a Fractured Paradigm', *Journal of Communication*, vol. 43, no. 4, pp. 51–58, Dec. 1993.
- [30] M. Haselmayer, M. Wagner, and T. M. Meyer, 'Partisan Bias in Message Selection: Media Gatekeeping of Party Press Releases', *Political Communication*, vol. 34, no. 3, pp. 367–384, Jul. 2017.
- [31] V. Larcinese, R. Puglisi, and J. M. Snyder, *Partisan bias in economic news: evidence on the agenda- setting behavior of U.S. newspapers*. London: Suntory and Toyota International Centres for Economics and Related Disciplines, 2008.
- [32] J. Cohen cohen@bc. ed., Y. Ding, C. Lesage, and H. Stolowy, 'Media Bias and the Persistence of the Expectation Gap: An Analysis of Press Articles on Corporate Fraud', *Journal of Business Ethics*, vol. 144, no. 3, pp. 637–659, Sep. 2017.
- [33] A. C. Gunther, 'Biased Press or Biased Public? Attitudes Toward Media Coverage of Social Groups', *The Public Opinion Quarterly*, vol. 56, no. 2, pp. 147–167, Jul. 1992.
- [34] S. Lyengar and K. S. Hahn, 'Red Media, Blue Media: Evidence of Ideological Selectivity in Media Use', *Journal of Communication*, vol. 59, no. 1, pp. 19–39, 2009.
- [35] D. C. Funder and S. G. West, 'Concensus, Self-Other Agreement, and Accuracy in Personality Judgment: An Introduction', *Journal of Personality*, vol. 61, no. 4, pp. 457–476, Dec. 1993.
- [36] Sendhil Mullainathan and Andrei Shleifer, 'The Market for News', *The American Economic Review*, vol. 95, no. 4, pp. 1031–1053, 2005.
- [37] E. Pariser, *The Filter Bubble: What The Internet Is Hiding From You*. Penguin UK, 2011.
- [38] J. Ehrlinger, T. Gilovich, and L. Ross, 'Peering Into the Bias Blind Spot: People's Assessments of Bias in Themselves and Others', *Pers Soc Psychol Bull*, vol. 31, no. 5, pp. 680–692, May 2005.
- [39] N. Geidner, I. Pjesivac, I. Imre, I. Coman, and D. Yuran, 'The Role of Interactive Graphics in Reducing Misperceptions in the Electorate', *Visual Communication Quarterly*, vol. 22, no. 3, pp. 133–145, Jul. 2015.
- [40] 1615 L. St NW, S. 800 Washington, and D. 20036 U.-419-4300 | M.-419-4349 | F.-419-4372 |
 M. Inquiries, 'Views of media bias', *Pew Research Center's Journalism Project*, 09-May-2017.
- [41] C. D. Urban, *Examining Our Credibility: Perspectives of the Public and the Press ; [report for the ASNE Journalism Credibility Project]*. American Soc. of Newspaper Ed., 1999.
- [42] D. Niven, *Tilt?: The Search for Media Bias*. Greenwood Publishing Group, 2002.
- [43] V. Larcinese, R. Puglisi, and J. M. Snyder Jr, 'Partisan bias in economic news: Evidence on the agenda-setting behavior of US newspapers', *Journal of Public Economics*, vol. 95, no. 9, pp. 1178–1189, 2011.

- [44] T. Groseclose and J. Milyo, 'A Measure of Media Bias', *The Quarterly Journal of Economics*, vol. 120, no. 4, pp. 1191–1237, Nov. 2005.
- [45] P. B. Lowry, D. W. Wilson, and W. L. Haig, 'A Picture is Worth a Thousand Words: Source Credibility Theory Applied to Logo and Website Design for Heightened Credibility and Consumer Trust', *International Journal of Human-Computer Interaction*, vol. 30, no. 1, pp. 63– 93, Jan. 2014.
- [46] B. Spillane, S. Lawless, and V. Wade, 'Perception of bias: the impact of user characteristics, website design and technical features', 2017, pp. 227–236.
- [47] S. Zwinger, J. Langer, and M. Zeiller, 'Acceptance and Usability of Interactive Infographics in Online Newspapers', in 2017 21st International Conference Information Visualisation (IV), 2017, pp. 176–181.
- [48] C. Chen and C. Chen, *Information visualization: beyond the horizon*, 2nd ed. London: Springer, 2004.
- [49] B. I. U. Dur, 'Analysis of Data Visualizations in Daily Newspapers in Terms of Graphic Design', *Procedia - Social and Behavioral Sciences*, vol. 51, pp. 278–283, Jan. 2012.
- [50] S. Pasternack and S. H. Utt, 'Reader Use & Understanding of Newspaper Infographics', *Newspaper Research Journal*, vol. 11, no. 2, pp. 28–41, Mar. 1990.
- [51] S. Bateman, R. L. Mandryk, C. Gutwin, A. Genest, D. McDine, and C. Brooks, 'Useful junk?', *Conference on Human Factors in Computing Systems Proceedings*, p. 2573, Apr. 2010.
- [52] E. Segel and J. Heer, 'Narrative Visualization: Telling Stories with Data', *IEEE Transactions on Visualization and Computer Graphics*, vol. 16, no. 6, pp. 1139–1148, Nov. 2010.
- [53] J. Vila and Y. Gomez, 'Extracting business information from graphs: An eye tracking experiment', *Journal of Business Research*, vol. 69, pp. 1741–1746, May 2016.
- [54] N. M. Urban and T. E. Fricker, 'A comparison of Latin hypercube and grid ensemble designs for the multivariate emulation of an Earth system model', *Computers & Geosciences*, vol. 36, no. 6, pp. 746–755, Jun. 2010.
- [55] 'https://prolific.ac/'. [Online]. Available: https://prolific.ac/. [Accessed: 22-Mar-2018].
- [56] 'http://www.gpower.hhu.de/'. [Online]. Available: http://www.gpower.hhu.de/. [Accessed: 22-Mar-2018].
- [57] 'dailymail.co.uk', Mail Online. [Online]. Available: http://www.dailymail.co.uk/home/index.html. [Accessed: 22-Mar-2018].
- [58] 'bbc.co.uk/news', BBC News. [Online]. Available: http://www.bbc.co.uk/news. [Accessed: 22-Mar-2018].
- [59] 'theguardian.com', the Guardian. [Online]. Available: http://www.theguardian.com/international. [Accessed: 22-Mar-2018].
- [60] 'YouGov | How left or right-wing are the UK's newspapers?', YouGov: What the world thinks. [Online]. Available: //yougov.co.uk/news/2017/03/07/how-left-or-right-wing-are-uksnewspapers/. [Accessed: 30-Jan-2018].
- [61] J. Cohen, *Statistical power analysis for the behavioral sciences*, 2nd ed. Hillsdale, N.J: L. Erlbaum Associates, 1988.

Appendix A1: Bias Ratings Data

The data in the tables in this Appendix shows the responses by participants in this survey when reporting the level of perceived bias in the images with which they were presented.

The data is presented in a number of columns, as follows:

	ID	The unique identification number of the participant.
	Path	The path through the experiment cube to which the participant was assigned.
Initial	Sample One	The first sample presented to the participant.
of Samples	Bias One	The bias rating given by the participant when presented with Sample One. This rating is on a continuous slider scale from 0 to 100.The slider defaulted to 0. Where a participant did not move the slider from the 0 mark, the response was recorded as NULL. This can therefore be taken to mean 0.
	Sample Two	The second sample presented to the participant.
	Bias Two	The bias rating given by the participant when presented with Sample Two. This rating is on a continuous slider scale from 0 to 100. The slider defaulted to 0. Where a participant did not move the slider from the 0 mark, the response was recorded as NULL. This can therefore be taken to mean 0.
	Sample Three	The third sample presented to the participant
	Bias Three	The bias rating given by the participant when presented with Sample Three. This rating is on a continuous slider scale from 0 to 100. The slider defaulted to 0. Where a participant did not move the slider from the 0 mark, the response was
		recorded as NULL. A response of NULL can therefore be taken to mean 0.
	Bias One	The bias rating given by the participant when presented with Samples One, Two, and Three side-by-side. The

Comparative	Bias Two	slider value defaulted to the response the participant had
Re-		given when originally presented with each sample. If the
Evaluations		participant did not change their rating for the bias of an
	Bias Three	article, the response was recorded as NULL. A response of
		NULL can therefore be taken to indicate that the value of
		bias is the same as in the original bias column.
Most Biased		The sample, from the three presented simultaneously to
		the participant, which they consider to be most biased.
		The participants could not view their previous bias ratings
		for those samples at this time. If a participant did not
		select a sample as the most biased, this is recorded as
		blank cell. Note that the result of this test may indicate
		that the participant recorded a different result to that
		when the samples were presented individually at the start
		of the survey.

								Comparative			Most
		InitiaLEva	luation	of Samples				Re-eva	luation		Biased
		Sampla	Biac	Samplo	Riac	Sampla	Riac	Riac	Riac	Riac	
ID	Path	One	One	Two	Two	Three	Three	One	Two	Three	
1	8	W1A0D2	37	W0A2D1	55	W2A1D0	46	NULL	NULL	NULL	W1A0D2
2	4	W2A0D2	1	W1A2D1	51	W0A1D0	69	NULL	NULL	NULL	W0A1D0
3	5	W2A0D0	52	W1A2D2	NULL	W0A1D1	10	NULL	NULL	NULL	W1A2D2
4	9	W1A0D0	NULL	W2A1D1	10	W0A2D2	NULL	NULL	NULL	NULL	W2A1D1
5	6	W1A2D0	58	W2A0D1	20	W0A1D2	75	NULL	NULL	NULL	W0A1D2
6	8	W1A0D2	NULL	W0A2D1	11	W2A1D0	80	22	NULL	NULL	W2A1D0
7	2	W2A2D0	5	W1A1D2	59	W0A0D1	96	NULL	NULL	NULL	W0A0D1
8	4	W2A0D2	48	W1A2D1	55	W0A1D0	64	NULL	NULL	NULL	W0A1D0
9	1	W2A2D2	14	W1A1D1	40	W0A0D0	4	NULL	NULL	9	W1A1D1
10	5	W1A2D2	65	W0A1D1	26	W2A0D0	NULL	NULL	NULL	NULL	W1A2D2
11	2	W1A1D2	77	W0A0D1	78	W2A2D0	82	NULL	NULL	NULL	
12	2	W1A1D2	24	W0A0D1	40	W2A2D0	8	NULL	NULL	NULL	W0A0D1
13	4	W1A2D1	90	W0A1D0	100	W2A0D2	61	100	NULL	56	W1A2D1
14	2	W1A1D2	5	W2A2D0	NULL	W0A0D1	5	NULL	NULL	NULL	W2A2D0
15	3	W1A1D0	12	W0A0D2	66	W2A2D1	21	NULL	NULL	NULL	W0A0D2
16	8	W1A0D2	78	W2A1D0	77	W0A2D1	80	NULL	NULL	NULL	W1A0D2
17	2	W1A1D2	10	W2A2D0	25	W0A0D1	5	NULL	NULL	NULL	W0A0D1
18	1	W0A0D0	NULL	W1A1D1	0	W2A2D2	0	NULL	NULL	NULL	W2A2D2
19	2	W1A1D2	10	W2A2D0	NULL	W0A0D1	30	15	NULL	NULL	W0A0D1
20	2	W0A0D1	NULL	W2A2D0	NULL	W1A1D2	25	NULL	NULL	NULL	W1A1D2
21	1	W2A2D2	0	W1A1D1	10	W0A0D0	0	NULL	NULL	NULL	W1A1D1
22	4	W0A1D0	92	W1A2D1	30	W2A0D2	23	NULL	NULL	19	W0A1D0
23	7	W2A1D2	25	W0A2D0	1	W1A0D1	74	NULL	NULL	NULL	W1A0D1
24	1	W2A2D2	0	W1A1D1	65	W0A0D0	75	NULL	NULL	NULL	W0A0D0
25	2	W0A0D1	70	W2A2D0	15	W1A1D2	40	NULL	NULL	50	W0A0D1
26	1	W0A0D0	9	W2A2D2	26	W1A1D1	1	NULL	NULL	NULL	W0A0D0
27	1	W0A0D0	52	W2A2D2	5	W1A1D1	92	75	1	NULL	W1A1D1
28	5	W0A1D1	100	W2A0D0	52	W1A2D2	40	NULL	NULL	NULL	W0A1D1
29	9	W0A2D2	9	W1A0D0	12	W2A1D1	25	NULL	NULL	40	W2A1D1
30	1	W0A0D0	21	W2A2D2	12	W1A1D1	5	NULL	NULL	NULL	W0A0D0
31	5	W1A2D2	NULL	W0A1D1	NULL	W2A0D0	NULL	NULL	NULL	NULL	W1A2D2
32	9	W2A1D1	58	W0A2D2	8	W1A0D0	2	NULL	NULL	NULL	W2A1D1
33	2	W0A0D1	77	W2A2D0	39	W1A1D2	46	NULL	NULL	NULL	W0A0D1
34	2	W2A2D0	19	W1A1D2	30	W0A0D1	34	NULL	NULL	NULL	W0A0D1
35	4	W2A0D2	20	W1A2D1	60	W0A1D0	39	NULL	NULL	NULL	W1A2D1
36	5	W1A2D2	69	W0A1D1	46	W2A0D0	16	NULL	NULL	NULL	W1A2D2
37	1	W1A1D1	42	W0A0D0	35	W2A2D2	4	NULL	NULL	NULL	W0A0D0
38	8	W2A1D0	19	W0A2D1	19	W1A0D2	51	NULL	NULL	NULL	
39	8	W2A1D0	30	W1A0D2	74	W0A2D1	48	NULL	NULL	NULL	W0A2D1
40	6	W1A2D0	12	W0A1D2	10	W2A0D1	8	NULL	NULL	NULL	W1A2D0

								Compa	arative		Most
		InitiaLEva	luation	of Samples				Re-eva	luation		Biased
ID	Path	Sample One	Bias One	Sample Two	Bias Two	Sample Three	Bias Three	Bias One	Bias Two	Bias Three	
41	2	W0A0D1	50	W1A1D2	100	W2A2D0	11	NULL	NULL	NULL	W1A1D2
42	7	W1A0D1	34	W0A2D0	23	W2A1D2	58	NULL	NULL	38	W2A1D2
43	7	W1A0D1	5	W0A2D0	36	W2A1D2	20	NULL	NULL	NULL	W0A2D0
44	1	W2A2D0	58	W1A1D2	24	W0A0D1	70	NULL	NULL	NULL	W0A0D1
45	6	W2A0D1	41	W1A2D0	9	W0A1D2	72	NULL	NULL	NULL	W0A1D2
46	6	W0A1D2	7	W2A0D1	1	W1A2D0	40	NULL	NULL	NULL	W1A2D0
47	4	W0A1D0	34	W1A2D1	58	W2A0D2	68	NULL	NULL	NULL	W2A0D2
48	2	W0A0D1	84	W2A2D0	16	W1A1D2	74	88	NULL	70	W0A0D1
49	3	W0A0D2	47	W1A1D0	NULL	W2A2D1	NULL	NULL	NULL	NULL	W0A0D2
50	1	W2A2D2	0	W0A0D0	30	W1A1D1	25	NULL	NULL	NULL	W0A0D0
51	6	W2A0D1	NULL	W0A1D2	25	W1A2D0	35	NULL	35	25	W0A1D2
52	4	W0A1D0	70	W2A0D2	5	W1A2D1	60	NULL	NULL	NULL	W0A1D0
53	2	W0A0D1	83	W2A2D0	100	W1A1D2	50	100	95	80	W2A2D0
54	2	W0A1D1	NULL	W1A2D2	NULL	W2A0D0	NULL	NULL	NULL	NULL	W1A2D2
55	7	W0A2D0	26	W1A0D1	50	W2A1D2	51	NULL	NULL	NULL	W1A0D1
56	6	W2A0D1	0	W1A2D0	0	W0A1D2	NULL	NULL	NULL	NULL	W1A2D0
57	5	W0A1D1	20	W1A2D2	20	W2A0D0	85	NULL	NULL	NULL	W2A0D0
58	6	W0A1D2	25	W2A0D1	15	W1A2D0	30	NULL	NULL	NULL	W1A2D0
59	9	W1A0D0	76	W0A2D2	9	W2A1D1	17	NULL	NULL	NULL	W1A0D0
60	4	W2A0D2	13	W1A2D1	11	W0A1D0	14	NULL	NULL	NULL	W1A2D1
61	3	W1A1D0	20	W2A2D1	20	W0A0D2	35	NULL	NULL	NULL	W0A0D2
62	7	W0A2D0	15	W2A1D2	95	W1A0D1	27	NULL	NULL	NULL	W2A1D2
63	3	W0A0D2	26	W1A1D0	78	W2A2D1	16	NULL	79	NULL	W1A1D0
64	1	W1A1D1	41	W0A0D0	74	W2A2D2	18	NULL	NULL	NULL	W2A2D2
65	1	W2A2D2	18	W0A0D0	49	W1A1D1	80	NULL	NULL	NULL	W1A1D1
66	5	W2A0D0	0	W0A1D1	31	W1A2D2	0	NULL	NULL	NULL	W0A1D1
67	1	W2A2D2	0	W0A0D0	82	W1A1D1	0	NULL	NULL	NULL	W0A0D0
68	8	W0A2D1	1	W1A0D2	6	W2A1D0	67	NULL	NULL	NULL	W2A1D0
69	5	W2A0D0	13	W0A1D1	NULL	W1A2D2	15	NULL	NULL	NULL	W1A2D2
70	8	W2A1D0	51	W0A2D1	0	W1A0D2	60	NULL	NULL	NULL	W2A1D0
71	6	W2A0D1	46	W1A2D0	30	W0A1D2	1	NULL	NULL	NULL	W2A0D1
72	8	W1A0D2	6	W0A2D1	NULL	W2A1D0	11	NULL	NULL	NULL	W2A1D0
73	9	W1A0D0	NULL	W2A1D1	50	W0A2D2	NULL	NULL	NULL	NULL	W2A1D1
74	3	W2A2D1	30	W0A0D2	10	W1A1D0	1	NULL	NULL	NULL	W0A0D2
75	9	W0A2D2	18	W1A0D0	NULL	W2A1D1	42	NULL	NULL	NULL	W2A1D1
76	7	W0A2D0	26	W1A0D1	50	W2A1D2	32	NULL	NULL	NULL	W1A0D1
77	3	W1A1D0	3	W2A2D1	3	W0A0D2	19	NULL	NULL	NULL	W0A0D2
78	3	W2A2D1	91	W0A0D2	86	W1A1D0	77	NULL	NULL	NULL	W0A0D2
79	5	W1A2D2	84	W0A1D1	9	W2A0D0	94	NULL	NULL	NULL	W1A2D2
80	4	W2A0D2	62	W1A2D1	24	W0A1D0	21	NULL	NULL	NULL	W0A1D0

								Compa	arative		Most
		InitiaLEva	luation	of Samples			[Re-eva	luation		Biased
ID	Path	Sample One	Bias One	Sample Two	Bias Two	Sample Three	Bias Three	Bias One	Bias Two	Bias Three	
81	4	W1A2D1	50	W0A1D0	46	W2A0D2	33	31	26	25	W1A2D1
82	5	W2A0D0	4	W1A2D2	4	W0A1D1	10	NULL	NULL	NULL	
83	9	W1A0D0	88	W0A2D2	0	W2A1D1	NULL	NULL	NULL	NULL	W1A0D0
84	4	W2A0D2	4	W1A2D1	5	W0A1D0	7	NULL	NULL	NULL	
85	3	W2A2D1	84	W1A1D0	82	W0A0D2	75	NULL	NULL	NULL	W1A1D0
86	7	W0A2D0	21	W1A0D1	59	W2A1D2	50	NULL	NULL	NULL	W1A0D1
87	7	W0A2D0	15	W1A0D1	10	W2A1D2	15	NULL	NULL	NULL	W0A2D0
88	1	W0A0D0	77	W1A1D1	85	W2A2D2	10	47	89	5	W1A1D1
89	9	W1A0D0	76	W2A1D1	83	W0A2D2	91	NULL	NULL	NULL	W2A1D1
90	3	W0A0D2	55	W2A2D1	10	W1A1D0	28	NULL	NULL	NULL	W0A0D2
91	6	W0A1D2	100	W1A2D0	100	W2A0D1	100	NULL	NULL	NULL	W0A1D2
92	3	W2A2D1	5	W0A0D2	57	W1A1D0	10	NULL	NULL	NULL	W0A0D2
93	3	W1A1D0	36	W2A2D1	14	W0A0D2	25	NULL	24	35	W1A1D0
94	9	W2A1D1	79	W1A0D0	86	W0A2D2	77	NULL	NULL	NULL	W1A0D0
95	6	W2A0D1	35	W0A1D2	65	W1A2D0	49	NULL	NULL	NULL	W0A1D2
96	9	W1A0D0	79	W2A1D1	62	W0A2D2	91	NULL	NULL	NULL	W0A2D2
97	6	W0A1D2	34	W1A2D0	45	W2A0D1	40	NULL	NULL	NULL	W1A2D0
98	6	W2A0D1	19	W1A2D0	70	W0A1D2	38	54	85	26	W1A2D0
99	4	W1A2D1	35	W2A0D2	18	W0A1D0	60	NULL	NULL	NULL	W0A1D0
100	4	W1A2D1	35	W2A0D2	45	W0A1D0	60	NULL	NULL	NULL	W0A1D0
101	4	W2A0D2	12	W0A1D0	44	W1A2D1	60	NULL	NULL	NULL	W1A2D1
102	3	W2A2D1	3	W0A0D2	26	W1A1D0	6	NULL	36	NULL	W0A0D2
103	4	W1A2D1	100	W2A0D2	75	W0A1D0	51	NULL	NULL	NULL	W1A2D1
104	4	W1A2D1	100	W2A0D2	2	W0A1D0	24	NULL	NULL	NULL	W1A2D1
105	5	W2A0D0	55	W1A2D2	65	W0A1D1	84	30	50	NULL	W0A1D1
106	7	W0A2D0	15	W2A1D2	20	W1A0D1	15	NULL	NULL	NULL	W0A2D0
107	4	W2A0D2	2	W1A2D1	20	W0A1D0	5	NULL	NULL	NULL	W1A2D1
108	9	W2A1D1	48	W1A0D0	38	W0A2D2	54	NULL	NULL	NULL	W0A2D2
109	8	W1A0D2	4	W2A1D0	6	W0A2D1	87	NULL	NULL	NULL	W0A2D1
110	7	W0A2D0	5	W2A1D2	100	W1A0D1	13	NULL	NULL	NULL	W2A1D2
111	5	W0A1D1	NULL	W1A2D2	20	W2A0D0	NULL	NULL	NULL	NULL	W1A2D2
112	7	W2A1D2	10	W1A0D1	NULL	W0A2D0	15	NULL	NULL	NULL	W0A2D0
113	6	W2A0D1	7	W0A1D2	5	W1A2D0	30	NULL	NULL	NULL	W1A2D0
114	5	W2A0D0	38	W1A2D2	64	W0A1D1	12	NULL	74	NULL	W1A2D2
115	5	W2A0D0	36	W0A1D1	73	W1A2D2	84	NULL	75	NULL	W1A2D2
116	8	W2A1D0	81	W1A0D2	30	W0A2D1	7	NULL	NULL	NULL	W2A1D0
117	3	W1A1D0	15	W0A0D2	29	W2A2D1	NULL	21	41	17	W0A0D2
118	7	W1A0D1	NULL	W2A1D2	3	W0A2D0	NULL	NULL	NULL	NULL	W2A1D2
119	5	W1A2D2	78	W2A0D0	0	W0A1D1	NULL	NULL	NULL	NULL	W1A2D2
120	8	W1A0D2	24	W0A2D1	0	W2A1D0	75	NULL	NULL	NULL	W2A1D0

		InitiaLEvaluation of Samples							Comparative Re-evaluation		
ID	Path	Sample One	Bias One	Sample Two	Bias Two	Sample Three	Bias Three	Bias One	Bias Two	Bias Three	
121	3	W1A1D0	13	W2A2D1	23	W0A0D2	70	7	49	NULL	W0A0D2
122	8	W1A0D2	42	W2A1D0	58	W0A2D1	24	NULL	NULL	NULL	W2A1D0
123	9	W2A1D1	85	W1A0D0	66	W0A2D2	77	NULL	NULL	NULL	W2A1D1
124	6	W2A0D1	NULL	W1A2D0	3	W0A1D2	NULL	NULL	NULL	NULL	W1A2D0

Appendix A2: Survey Responses

The qualitative data collected for this survey, consisted primarily of responses to multiple choice questions. Some questions were posed before the survey; others were posed after the survey.

In addition to multiple choice questions, each participant also provided his or her date of birth (DOB).

A unique identifier (ID) was assigned to each participant.

Each question is assigned a letter (A to K). The following is the list of question and the permitted answers. Questions A to H were answered prior to the survey. Questions I to K were answered following the survey.

	Question	Answer codes
Α	"What gender are	0 = "Male"
	you?"	1 = "Female"
		2 = "Other"
		3 = "I prefer not to say"
В	"How long do you	0 = "Less than one hour a day"
	spend on the Internet each day?"	1 = "1 – 2 hours a day"
		2 = "2 – 4 hours a day"
		3 = "4 – 6 hours a day"
		4 = "6 – 8 hour a day"
		5 = "8+ hours a day"
С	"What is the	1 = "Elementary school only"
	highest level of education you have	2 = "Some high school, but did not finish"
	completed?"	3 = "Completed high school"
		4 = "Some college, but did not finish"
		5 = "Two-year college degree / A.A / A.S."
		6 = "Four-year college degree / B.A. / B.S."
		7 = "Some graduate study"
		8 = "Completed Masters or professional degree"
		9 = "Advanced Graduate study or Ph.D."
		10 = "Other"

	Question	Answer codes
D	"What is your	1 = "Under \$10,000"
	range for 2018?"	2 = "\$10,000 - \$19,999"
		3 = "\$20,000 - \$29,999"
		4 = "\$30,000 - \$39,999"
		5 = "\$40,000 - \$49,999"
		6 = "\$50,000 - \$59,999"
		7 = "\$60,000 - \$69,999"
		8 = "\$70,000 - \$79,999"
		9 = "\$80,000 - \$89,999"
		10 = "\$90,000 - \$99,999"
		11 = "\$100,000 - \$109,999"
		12 = "\$110,000 - \$119,999"
		13 = "\$120,000 - \$129,999"
		14 = "\$130,000 - \$139,999"
		15 = "\$140,000 - \$149,999"
		16 = "\$150,000 - \$159,999"
		17 = "\$160,000 - \$169,999"
		18 = "\$170,000 - \$179,999"
		19 = "\$180,000 - \$189,999"
		20 = "\$190,000 - \$199,999"
		21 = "\$200,000 - \$209,999"
		22 = "\$210,000 - \$219,999"
		23 = "\$220,000 - \$229,999"
		24 = "\$230,000 - \$239,999"
		25 = "\$240,000 - \$249,999"
		26 = "Over \$250,000"

Question	Answer codes
"Which best	1 = "Stay at home Parent"
current	2 = "Student"
occupation?"	3 = "Unemployed"
	4 = "Professional"
	5 = "Semi-professional"
	6 = "Skilled professional"
	7 = "Skilled manual"
	8 = "Semi-skilled manual"
	9 = "Unskilled manual"
	10 = "Other"
"What most closely	0 = "Strong Liberal"
describes your political	1 = "Not so strong Liberal"
viewpoint?"	2 = "Independent leaning Liberal"
	3 = "Independent"
	4 = "Independent leaning Conservative"
	5 = "Not so strong Conservative"
	6 = "Strong Conservative"
	7 = "Other"
	8 = "Don't know"
"Which mediums	1 = "Radio"
do you access News from most	2 = "TV"
often? Select all that apply."	3 = "Newspaper"
	4 = "Internet"
	5 = "Social Media"
	6 = "None"
	7 = "Other"
	Question "Which best describes your current occupation?" "What most closely describes your political viewpoint?" "Which mediums do you access News from most often? Select all that apply."

	Question	Answer codes
н	"If the Internet is	0 = "Less than once per day"
	primary sources of	1 = "1 - 2 times per day"
	News, how many	2 = "2 - 4 times per day"
	it each day?"	3 = "4 - 6 times per day"
		4 = "6 - 8 times per day"
		5 = "8 - 10 times per day"
		6 = "10 + times a day"
I	"Did you recognize	0 = "None"
	any of the below News websites	1 = "The Guardian"
	used in this	2 = "The Telegraph"
	experiment? Select each that you	3 = "The Independent"
	recognized."	4 = "The Daily Mail"
		5 = "The Spectator"
		6 = "New Statesman"
		7 = "Al Jazeera"
		8 = "BBC"
		9 = "Reuters"
J	"Do you consider	0 = "None"
	any of these News websites used in	1 = "The Guardian"
	this experiment to	2 = "The Telegraph"
	be especially biased? Select each	3 = "The Independent"
	that you consider	4 = "The Daily Mail"
	biasea.	5 = "The Spectator"
		6 = "New Statesman"
		7 = "Al Jazeera"
		8 = "BBC"
		9 = "Reuters"

	Question	Answer codes
К	"When judging bias in news articles online, which of the following do you consider to be significant elements of bias? Please select all	 Answer codes 0 = "Lack of balance" 1 = "Unfairness" 2 = "Subjectivity" 3 = "Distortion affecting one side of a story" 4 = "Not giving one side of a story as much coverage as the other" 5 = "Expression of personal opinions by a journalist"
	τηστ αρριγ.	 6 = "Providing sources for one side of a story and not the other" 7 = "Providing data to support one side of a story and not the other" 8 = "Order in which sides of a story are reported" 9 = "Other"

The data in the following table shows the responses by participants to survey questions before and after rating the bias of the articles with which they were presented, as detailed in Chapter 3.

	Responses to	Responses to pre-bias rating survey questions							Responses to reflective questions			
ID	D.O.B.	Α	В	С	D	Ε	F	G	Н	1	J	К
1	25/05/1991	1	4	6	1	5	2	1,2,3,4,5	3	0	0	1,2,3,7,8
2	17/09/1953	0	2	6	7	3	1	3,4	6	1,2,3,4,7,8,9	4,7	0,3,5,6
3	08/03/1986	0	3	6	7	6	2	1,2,4,5	2	0	0	0,1,3,5,7
4	09/06/1999	1	3	3	1	2	2	2,4,5	1	0	0	2,4,5
5	01/04/1987	1	4	9	10	4	0	4,5	1	1,2,8	0	0,3,5,6,7
6	06/08/1992	1	1	6	6	6	1	4	2	1	1	4,7
7	04/09/1994	0	5	6	6	6	2	4,5	6	1,8	4	0,3,4
8	13/06/1988	1	1	5	3	1	0	2,4	1	0	0	3
9	01/04/1989	0	5	3	2	8	1	4,5	1	1,4	4	2,3,5
10	30/10/1991	0	1	5	4	2	5	4,5	4	8	4	0,2,4,5,7
11	15/10/1980	1	4	8	16	4	0	3,4,5	2	1,2,3,4,7,8	1,2,3,4,6,7	1,2,4
12	31/12/1988	1	3	4	2	2	3	2,4,5	5	4	0	0,1,2,3,4,5,6,7
13	17/11/1995	0	5	4	6	3	2	4,5	1	4	4	3,4,5,7
14	03/04/1989	1	4	6	6	5	1	1,4,5	3	2,4	0	1,3,4,7
15	15/02/1988	1	3	6	4	4	0	3,4,5	1	2	3,4,6,7,9	0,3,4,5,6,7
16	15/10/1980	0	4	8	16	4	0	3,4,5	2	1,2,3,4,5,6,7,8	1,2,3,4,5,7	1,2,3,4
17	01/02/1986	0	4	4	3	5	4	4,5	2	1,4	0	4
18	27/10/1964	1	3	6	4	4	2	2,3	2	1,2,3,4,7,8,9	0	0,1,3,5,6,7
19	24/12/1980	1	3	4	4	5	2	4,5	2	0	1,5,7	2,5,6,9
20	26/02/1998	0	3	3	1	2	3	1,4,5	4	0	4,7	0,1,2,3,4,5
21	19/01/1976	1	5	6	10	6	0	4,5	6	4	4	0,1,2,3,4,5,6,7,8,9
22	06/04/2000	0	2	2	1	2	2	4,5,7	1	1,3,8,9	3,9	1,2,3,6,7
23	18/11/1991	0	5	4	1	2	0	2,4	2	0	4	0,2,5
24	04/04/1981	0	3	8	8	4	5	1,4	6	0	0	0,3,4,6,7
25	18/01/1998	0	5	4	1	2	3	4	0	0	1	0,4
26	01/01/1999	0	3	5	1	2	3	3,4	1	4	4	0,1,2,4,6,7
27	25/04/1977	1	2	3	3	9	3	4	1	0	0	2,5,8
28	13/10/1984	0	5	6	8	4	5	1,2,3,4,5	3	0	0	0,5,6,7,8
29	24/11/1986	0	3	6	9	4	3	4	2	0	0	0,1,6,7
30	01/08/1973	0	4	3	10	5	0	4	6	8	0	0,3,4,6,7
31	05/05/1972	3	2	6	1	5	8	2,4	3	4	0	5,7
32	16/10/1998	0	5	4	2	1	8	4,5	6	0	0	3,5
33	11/11/1978	0	5	6	8	4	6	2,4,5	6	1	1,2,4,7,8,9	0,1,2,3,4,5,6,7,8
34	12/07/1970	0	5	6	11	6	4	2,5	1	0	0	7
35	10/06/1954	1	2	5	4	10	1	2,4,5	3	0	0	3
36	01/04/1977	1	2	4	2	9	0	1,4,5	1	0	4	0,1,3,7
37	24/08/1972	0	1	6	5	6	3	4	1	1,3,4,7,8,9	4,9	0,1,2,3,4,5,6,7
38	18/05/1952	0	2	8	5	10	6	1,4	1	1,2,4	5,6,7	3,4,6
39	18/02/1981	0	3	7	6	6	1	4	1	0	0	2,3,6,7
40	05/07/1986	0	3	6	7	4	0	4,5	2	0	4,7	0,1,2,3,4,5,6,7,8
41	22/03/1985	0	5	6	8	5	4	2,4,5	4	0	4,7	0,1,2,3,5,6,7
42	12/09/1975	0	3	8	18	4	5	2,3,4,5	2	1,2,3,4,8	1,3,4	4,5,7
43	11/11/1999	0	5	3	1	2	2	1,4	0	1,4,8	4	7

	Responses to pre-bias rating survey questions F						/ que	Responses to reflective questions				
ID	D.O.B.	Α	В	С	D	Ε	F	G	н	1	J	К
44	03/03/1996	0	5	4	2	2	3	4	3	0	0	0,2,4,6
45	14/04/1983	0	4	4	5	8	3	2	1	8	0	0,2
46	28/09/1989	0	5	4	2	2	2	4	4	1,2,3,4,7,8,9	1	2,3,5,7
47	03/03/1996	0	5	4	2	2	3	4	3	0	0	0,1,2,4,6
48	03/04/1985	1	4	4	1	3	2	1	0	0	0	1,2,3,4,5,6,7,8
49	06/06/1984	1	5	7	4	6	0	2,4,5	1	1,7,8,9	2,4,5	2,3,4,5,6,7
50	22/10/1978	1	5	4	1	2	3	6	0	8	0	0,1,2,3,4,5,7
51	16/01/1970	0	3	6	18	4	0	1,4	4	0	0	0,2,3,4,7
											1,2,3,4,5,6,	
52	23/04/1971	0	5	5	1	3	3	4	3	0	7,8,9	1,2,3,5
53	19/03/1982	0	2	9	15	4	3	2,3,4,5	2	1,4,7,8	3,4,7,8	9
54	10/08/1983	1	5	6	9	6	0	4	6	0	0	1,2,3,5
55	27/05/1987	1	2	8	2	2	0	4,5	2	1,4,8	4	2,3,5,6,7
56	06/07/1977	1	2	3	1	1	6	6	4	0	0	9
57	14/05/1993	1	4	4	5	4	0	1,4,5	1	0	0	2,3,5,9
58	06/05/1956	1	1	5	2	10	1	2,3,4	0	0	1,2,4,9	0,3,4,5,6,7,8
59	02/05/1996	1	1	4	5	1	3	2,3,4,5	2	0	0	3,4,5
60	29/03/1983	0	5	6	3	4	5	1,2,3,4	3	0	7,9	2,3,5,7
61	07/10/1976	0	3	3	1	10	4	2,3,4,5	2	1,2,4	0	3,4
62	14/05/1993	1	4	4	5	4	0	1,4,5	1	0	4	2,3,5,9
63	26/02/1991	0	1	6	3	5	5	2,4	2	4	9	4,5
64	07/07/1988	1	3	5	3	5	0	2,4,5	6	8	0	5,7
65	25/12/1996	0	3	3	10	7	3	2,4	3	0	0	4
66	07/11/1988	0	2	6	2	4	3	4,5	1	0	4	0,2,5,6,7
67	09/05/1982	1	3	4	8	5	6	1,2,3,4	0	1,8,9	0	0,2,3,4,5,6,7
68	03/09/1982	0	5	3	2	8	2	4,5	6	1,2,4,7,8,9	4,7	0,3,4,8
69	16/02/1988	1	2	3	3	3	2	2,4,5	2	1,4,7,8	0	4,5,6,7
70	23/08/1981	1	3	4	2	6	0	4	1	1,2,3,4,7,8,9	1,4,7	0,1,2,3,4,5,6,7
71	25/07/1994	1	3	4	3	2	1	4,5	2	0	0	2,5
72	14/09/1984	1	5	5	5	1	4	1,4	0	0	0	0,1,2,4,5,6,7
73	12/12/1992	0	4	3	1	3	2	2,4,5	1	0	0	0,1,3,4,5,6,7
74	17/07/1981	0	2	8	8	4	2	4,5	2	4,8,9	1,4,7	2,3,4,5,6
75	27/04/1999	1	2	3	1	2	2	2,4	0	1,4,7,8,9	4	2,4,6,7
76	03/09/1985	1	5	6	6	6	0	4	1	1,2,3,4,7,8,9	0	0,2,3,4,5,6,7
77	04/03/1955	0	1	8	10	4	5	1,2,3,4	2	1,2,3,4,7,8,9	7,8	0,1,2,3,4,5,6,7,8
78	25/10/1977	0	4	8	26	4	6	1,5	4	2,4,6,8	2,4	5,7,8
79	12/04/1979	0	2	6	10	4	2	1,4	2	0	0	0,1,2,3,4,5,6,7
80	01/09/1963	0	3	5	2	2	3	1,4	2	0	0	4
81	08/07/1954	1	3	4	13	1	3	4	1	4,8	0	0,1,3,4,6,7,8
82	08/11/1963	0	5	6	6	4	0	4	1	1,9	0	1,3,5
83	11/11/1973	1	1	8	12	1	1	1,4	2	0	0	2,3,5
84	02/11/1956	0	2	8	8	4	5	2,3,4	2	2,3,4	0,4	3
85	25/10/1977	0	4	8	26	4	6	1,2,3,4,5	4	3,4,5,8	2,3,4,6,8	5,6,8

	Responses to	o pre	e-bia	s rat	ing s	urvey	/ que	estions		Responses to re	flective quest	ions
ID	D.O.B.	Α	В	С	D	Ε	F	G	Н	1	J	К
86	04/05/1988	0	2	5	5	4	3	4	3	8	8	4
87	11/12/1996	1	5	4	2	2	0	4,5	1	0	0	0,2,5,7
88	22/08/1977	0	3	7	5	6	2	1,4	2	1,2,3,4,6,7,8,9	1,2,4,8	0,2,3,5,6,7
89	25/10/1977	0	4	8	26	4	6	1,2,3,4,5	4	2,8	2,3,4,6,8	3,5,6,7
90	19/11/1981	1	5	4	2	1	0	4,5	3	1,4	1,4	0,1,2,3,4,5,6,7,8
91	16/07/1982	1	5	8	7	4	3	6	0	0	0	2
92	14/12/1989	0	5	6	8	4	0	2,4,5	6	4	4	0,2,4,5
93	27/08/1988	1	5	6	8	4	4	1,2,4,5	1	1	0	0,3,4,5,7
94	25/10/1977	0	4	8	26	4	6	1,2,3,5	4	2,3,4,8	2,4,8	3,6
95	11/03/1996	1	2	6	5	5	0	1,3,4	3	0	0	0,1,2,4,6,7
96	02/01/1964	1	5	8	8	6	3	4	6	8	4	3,4,5,6,7
97	01/10/1990	0	3	8	2	2	3	4	1	0	0	0,1,2,3,4,5,6,7,8
98	27/04/1966	0	2	5	3	10	6	1,2,4	2	8,9	1,2	0,1,2,3,5,6,8
99	14/08/1989	1	3	5	1	1	4	2,4	1	0	0	0,2,3,4,5,6,7,8
100	29/12/1987	0	4	8	1	3	2	4	3	0	2,4	3,6,7
101	01/10/1992	1	4	4	3	8	2	2,4,5	1	1	1,4	3,4,5
102	07/02/1995	0	4	6	2	2	0	2,4,5	3	1,3,4,8	0	0,1,2,3,5,7
103	20/12/1989	2	2	7	3	5	1	1,4	0	1,4,7,8,9	4	0,2,3,4,5,6,7
104	07/04/1960	1	4	8	3	6	5	4	1	1,3,4,7,8,9	4	0,2,3,4,5,6,7
105	01/10/1981	0	2	5	4	4	2	1,4,5	1	9	4	2,3,4,6,7
106	04/12/1963	0	2	4	8	6	0	1,4	3	0	1	0,1,2,3,4,5,6,7
107	18/02/1990	1	4	4	1	3	1	4,5	0	0	0	1,2,3,6,7
108	03/03/1968	0	3	5	8	6	1	4,5	4	1,4,7,8,9	1,4	1,2,4,6,7
109	07/01/1996	0	3	4	2	10	3	2,4,5	0	0	0	5,6,7
110	01/04/1988	0	5	4	2	4	6	1,4,5	1	1,2,3,4,7,8,9	1,2,3,4,8	0,1,2,3,4,5,6,7,8
111	22/05/1990	0	2	6	7	4	2	4,7	3	1,2,3,4,7,8,9	4	0,1,2,3,4,5
112	16/04/1986	0	2	4	4	7	1	4,5	1	0	0	2,4,5,7
113	01/06/1968	0	2	6	7	6	5	2,4,5	1	1,4,5,7,8,9	1,7	1,3,7
114	11/11/1990	1	1	6	6	4	1	1,4	1	0	0	2,4
115	15/03/1952	0	3	8	3	10	3	4,5	4	4	4	1,2,3,4,5,6,7
116	11/03/1989	0	3	4	1	1	2	1,4,5	6	8	0	0,1,2,3,4,5
117	01/03/1997	1	4	3	2	10	2	3,4,7	3	1,2,3,4,7,8,9	4,7	0,1,2,3,4,5,6,7,8
118	12/07/1988	1	2	5	1	1	3	1,3,4,5	1	1,2,3,4,8	0	3,4,5,7
119	19/07/1960	0	2	6	11	4	6	1,2,4	1	0	0	0,1,2,3,4,5,6,7,8
120	23/06/1976	0	3	4	6	4	0	4	1	1,4,8,9	0	2,3
121	25/07/1983	0	2	8	12	4	1	1,2,3,4,5	3	0	4	0,1,2,3,4,5,6,7
122	07/02/1990	1	5	5	2	5	0	4	1	0	4,5,8	0,3,4
123	08/08/1979	0	5	3	4	4	2	4	5	0	0	0,1,2,3,4,5,6,7,8
124	01/07/1966	0	5	6	10	4	5	1,2,3,4	1	1,2,3,4,7,8,9	0	0,1,3,4,6,7

Appendix A3: Participant Information

Experiment - Bias Perception in Online News

Participant Information Sheet

Thank you for your interest in taking part in this task-based evaluation. The information below is provided to inform your decision about consenting to take part in this experiment.

Details of the Experiment

The aim of this experiment is to ascertain how graphics accompanying news articles affect the perception of the articles' bias.

The experiment will begin by asking participants to fill out some basic profile information. This will include questions on:

- News access habits
- · Political ideology questions

Participants will then be asked to rate the Bias of various new website articles. After rating 3 such articles, each participant will be asked to select the most and least Biased and to provide additional information as to why they rated them as such.

Participants will then be asked reflective questions on the Bias of various news mediums, what formats they consider to be Biased and which formats they consider the least Biased.

The results of this experiment will form part of the research being undertaken as part of a structured Masters degree in Computer Engineering at the School of Computer Science and Statistics, Trinity College Dublin. The results may be published at an international conference or journal. This publication will ensure that participants' identities will not be revealed.

Benefits of this Research

This research is being undertaken to investigate how graphics accompanying news articles may affect the perceived bias of those articles.

The experiment uses websites, news articles and images from mainstream news websites. In most cases, they have been heavily adapted to suit the experiment. Copyright is held by the originators for the original versions of the websites, articles and images. Copyright on the modified versions of each is not claimed by the organisers of the experiment. Use of the original articles and modified versions is done under Fair Use. No financial reward is being earned from the original or modified content. None of the content, websites, articles or images, original or modified will be distributed outside of the experiment.

Cian Flynn Trinity College Dublin Email: flynnc16@tcd.ie

Bias Experiment

Crowdsourced experiment of users' perception of Bias in different News Articles



© Cian Flynn, Trinity College Dublin 2018. Email: Cian Flynn.

Appendix A4: News Articles and Distortions Used in the Experiment

The following images show the websites presented to participants in the experiment for their evaluation of perceived level of bias.

The label associated with each website corresponds to the website (W), story (A), and level of distortion (D), as discussed in Chapter 3.

W0A0D0

W0A0D1

W0A0D2



W0A1D0

W0A1D1

W0A1D2



xxxii

W0A2D0



UK life expectancy increases slightly, although progress slows	Res Gins Cale van wordt 🔠
REPELLINES AS 21, 27 Sequences (SU2) LIPEATES II (N. 27 Sequencies 2007	
🛉 Stans 🗾 🖗 🎦 🔛 🛋	
Life associately in the UK has increased slightly for both menund women - alteragin encourse is dowing more forces along	DON'T MISS
4 girl seen harawar 2004 and 2006 a networked in few unit 1020 years risk, whereas a links tory would seen 2023 years, articipassas of 0.554 yea 2021 10 2020, the OF iso fer- baricent Rudinics (DND) have st.	U.Stoch energy for the exclusive state SECONCS and SECONCS AND SEC
Life expectancy in the UK	* Dealth Products?
1 19	Anne destants
Women 79.2	· Diate and Delete
768 Men O	Artige Westgare, and de analysis of a same below the UNITS of the terrar part of a same
0	Access rates and second rates and out on second rates while takes and a rates
1982	P (tean bac service Basel the Land Josevice R III (ga attest Styl) R
Life capeciancy in the UK graphic	Nancionas Ant Donne
Make how is the same two year period have a 21% chalse of living until 80, while a tensive how is 30% characteristic coming a nanogenation. Currently a 65-year-old man on expect to one a hinter (85 years of 10+ whereas a university of the state has a nanogenetic to be a hinter (85 years).	SECT PERSON IN SECTOR PERSON IN ATTACK INTERNAL ADVICEMENT
Life expectancy by nation	Portania o faculto entinente la Societtaria Restance de cancella ser las consultarias de cancella ser esta consultaria de consultaria de consultaria esta consultaria de consultaria de consultaria esta consultaria de consultaria de consultaria de consultaria de consultaria esta consultaria de consultaria esta consultaria de consu
79.5 78.5 78.4 81.2	Pressure of the states
	Activities and a second and a s
England N Instand Wales Scotland S Ingenepundekinguros ero for 2014 to 2016 Life oppositiency in the UKby meson	Stars fire und sees Som Kab Oras at Transformers at Transformers
The latest inercase amounts to around one month more for more, and around a far tright more for women semialated with the processis year.	CENTRA SOCIETARE
ONS Buildean Bachiz Sunders and: "The rate of increase in Relationships in the UR has slowed in recent years.	Section Revi Interaction and Link Wards at a constitute Wards at a
COMB Company Comp	A line form them them whether a strength of the second strength of the strength of the strength of the second strength of the strength of the strength of the second strength of the strength of the strength of the strength of the second strength of the strength of the second strength of the strength of t
© 1 g too the set of t	Trate Lases Gooder Sectory (Dean For Accidence) and month of Sectory month of Sectory
prove he tools access to 76:2 years for matca and 82:0 years for fermices." Impleed alreaded the rights of life expectances with 83:1 years for access and 78.6 for many followed by Netthern behand, Barks and Secolard with the lowest on 81:2 for many and 79.1 for a new.	Marker Section 1
50 0010 V	Variation of the second second
A newtron tarty top could report to havin (0.2 press and a tarty of 82 4 page on (1.4 page 2014) 2 12 4 page on (1.4 page 2014)	Autoration to a Mathematical Strength and the Strength and the Strength of Strengthof Strength of Strength of Strength of Strength of Stre
⊘ 2 ^A / ₁₂ Soc 0495-010 Feadle ⊕	These Constant
The entire near two doesn't restrictly relations in the mean and period are reactained throughout the individual's file.	WAL JAN SCH
The figure core short a leading explanic warred entire this year that the experiment at birth was steprating.	A MA LY MA I THAN MARY Classifier Mars of the Markey A resource in the second
A main here in 2014 to 2016 has a 21% chance of survivor to at	
from the set of the contrast tasks in 22% , there is no $1/\sqrt{6}$ - 0.00 in 2.5 and 1.00 , 1.00 , 1.00	North Address of the Second
Unit 2 and Sufficients Transition (1997) Robusson 34 Michael Morrison, directors of the Instatus of Health Equilibre Unit-work Delega Lendon, calefort for an upget or commission inmit the subsections within the savial hard regionand to a health, and expressed expression games ing on health and used in an	Anni Frynan Preis yw add ganwad of Annian yr ydda Darnag y addau Darnag y addau Darnag y addau Arywad adau farinig
Figures show that the investe in fife expectance between 2010 2012 and 2014 2016 was classical bornman than hait, However, the Chill suggested the absences would be down to an earlication of efforts to improve the span, such as earling down smeking	
Package: 2015/2010 of an U.K. strate-signed 85 has an assessing 15.2 years of Mic remaining while a women has 20.0 years www.yething2014-bit 00.5 year 3 dec 20 and	
O.S. 5 See CRONSTATION 0	

W0A2D1

W0A2D2



W1A0D0

W1A0D1

W1A0D2





NEWS

na gas o tina de evene a Marcio dos das sal ar difor Sensarios (Evenenias S27 () de madero 27, es failor 75. Hilders view and even (2) in taken 7%.
Will you field the pitch title Christman?
Rosse prices rise functor in Cast Michaels.
Rosses prices rise functor in Cast Michaels.
Rosses prices rise functor in Cast Michaels. Non-well's relation contraining of the resident of the factors for marks of the Theodory Policy Contrains a Non-tag most space from day. Inflation rate hits 3.1% Consumer Prices Index (%)





N Taxing has said that is consider to basis to past the call rise to write to the character was in Se to the NSL in October Dire gas oorio Nevenbes. 2016, alar sel case alived lefter a litic publication for Sectory, actors for Sector I Deplevel and in standards influence Record

55



50212.375

Drag to the high control of information of the control of the second control of the particular devices the control of the particular devices of the device the second of the devices of the device of Whet has view larger, bough is the formed of the liver ways are a same unclease from every in a backway of page to may not a registration object from every in a backway of page to may not a registration object from every of a close on the type back from Eggen prior case had to tagget day, save introng has a sequence to charge tight prior to case. Accordingly, paids the second of Second edge with the second of the second of Second edge with the second of the seco Furthers used with the probability of the second se



10 miles



In PCF server the Construction and a structure of the Construct Polar Index of Pol structure of a structure of the Index of the Westge with server based and structure of the Westge with server of a structure of the structure of structure of the structure of Encourse in the separate Listen encourse a faither internet about 1 add does the made of the determine Policy Control that size day manches on Thingday.

Inflation rate hits 3.1%

industria (CRI) a

10 All Consol Vages - Elisar condetescent advanted



. a debad værstere



. .

Mit Dennes has an of the line expected strainers in power to deduce the forward of The set if no many expects the change in each if Decaration (2010), where the set where it of the intervalue theory on the set of the s Mr Samor's statt offer will be automodin Februhu whee will a will avoid a second biological fiber

AN A



In the day to receive at the 25 hours is which here they have the day is a ball for received received and existing and the day is a second of the day of weak to be applied by the day of Burnterson since deals ensure the appendixed stay. This interview with the lack is set-well here a source an interpretation of a star work in early the product star work and a source and the product and printing in the star work and the product star the City is well at the initial relation of the product at City part and year the City is well at the initial relation of the product at City part and year





W1A1D0

W1A1D1

NEWS

W1A1D2

NEWS



Nonaly of Hap to see 1 soldward y laws in some 1 Visi leaves from the Leaf straps and ward to the set of Life



Most Read Lenser differences 1 personale to enge 1 N a Boothern the correspond 2

Share this story a new





Advances white a new The UP - of the picture of the second secon en. Le vez directiv de cale peloc The the stated "We anoighed thoughting with Contyrest as well as a finite related and state entropy an activities between being any colors of registric terms taking a structure is encourage an appendix 5%". Inside a third in a reconcept or approximately a start in the test of the test

Related Topics

Share this story and an

B SNOTTHEFT

Most Read theredge Life agric service 1 booldaar is a cristof to indep Weis location the black bings 2 weise wet have together of UK

Co.

W1A2D0

Life expectancy rises "grinding to half" in England Concertor # F V Concertor And Participation Concertor # F V Concertor And Participation Concertor # F V Concertor And Participation

entering and a second sec

EXAMINET

Filling noses of the expectancy are gritiding to a help in England after more than 100 years of continuous programs, says a leading health-report.

Is austerily to blanne? Ho call has had to dow for consists about he case. By new primes, where you do camery he private ofe

Exception of accorr factors such as explorations, any electric factor web ing conditions comparedly of other deal this expediency by industring the type And as address was plotting processes on these, they may intuit to influencing thereign takes.

In a door 1 gridgenze what he so it was "interry" land og setternene for ne helfs en den en dans, skielt mend fre spel to it helfs in dan pergelerene til team mindemet and to stand after tilse itt septemet

The store specificate opposing more passe for the first law of people a fit determine, although the process index i may be be the to be the store in the level to off. Have we simply reached the outer limits of human life? Have we will have reached the outset limits of numeric IP/2 is there also well well well have made in the state of the state binaries and the state in the state of the state of the state binaries and the state is the state of the state of the state binaries and the state is the state of the state of the state binaries and the state of the state of the state of the state binaries and the state of the state of the state of the state of the state binaries and the state of the state of the state of the state of the state binaries and the state of the state of the state of the state of the state binaries and the state of the state the state of the state and the state of the state state of the state of th

initial values from the machinese series for the expectation ratio for the

The government must act! At each the design of the set of the set

*Do other carbins, the other precise of market parts of the definition of the inter-investigation has now missioned people with research with result entert. Descent prime and parts of the booth and entering to introduce they have been administered international entertainty.

nam an mening in mening and begin the set of plane and passes without " "I no gover instal has to and below the set of plane and passes without " I no incomposition and the dependences of the set of the mening install. It was related the mening was heard in sector of the set of

Our matcher was have to interest in the set of . The operative contrast in terms with overcreative make in a monthlash, which see any product was all and waters have. Anythis was characterized and when you're instantial sets. 2019 and one have contrast with the Stragget instantial sets. 2019 and one way instantial sets and the sets of the sets of the set of the sets of the memory instantial sets.

How do you improve your life expectancy? The next securities to behild Thi nexts take of an exercise system.

Brownie, inverso of new centrics and seconds, plays a key value of deer second ingood finally care when you can sola. Receive these development, seconds with of the LK for the matter have your waves or development for MAS. One of the places with the loggest gap in the second may via has been widely reported to many the directed Town Down in Standard to a second of many in The Electric Concerns weat conversally. I now a heating when the second se nemeng adardient, actions providence est painty. International an 2017 Action for one brane are politic reported, directories, as antifered also and the second sec

Rood more from Nick Follow Rick on Turber pe Ventrerarbe

Related Topics

Share this story -----

art diside a fact rate by it is size -

exectance in English is currently 65 for wants 1 and 75 + for more

NEWS

W1A2D1

4

1012

and and and and and and another and another an



Dany CH to be "Automat blancks projection (C) to be being real 2000. By blance, and has not well of the beginnermed on the Well bally Capital com-research to obside the masses is the second second welly received down 2010 in the second s England Produces son if the part of the UK. Measure (2011) we 2020, it is opposing a first instance of types your server the even for assess and by the services (3.5 services therein) have a supervision wave some (3.5 services therein) and have been any defer-tion on A2000.

1740-0-0-0

Features

■

We cannot be write real tool do not be a set of the set

and the second second

A later transfer frages and a frages

Host Read

EBCITIERS CONTRACTS 3

Extrational precision from 4

To visi licent-track 5 Degenties Nines of partici-bening hose

Paper I creek Graphe 6

Kind user Jours and 7 Memory faces is beined

Steam the City and Centrics 8 The CK and ED agree to ms to 9 Deset to ne for ported

Numeral Ageneration The Intern 10

From Around the Web

And Mid with raising describer TV provided statistics treatment (they ratio self-driving randown wher death)

A more consistent and a constraint with the second second



Arrive-Sector street investment Sector and an end that provide a Distribution of an end that provide a characteristic and that provide a Distribution of the sector at the Sector Sector Sector Sector at the Sector Sector Sector Sector at the Sector Sec Arvery College a Russell ex.
Elson signs predict forme
General Onion Mission Elson
School and Lincols
Elson Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson
Elson Septer Lorred Cospie 6 Tractical Internative Res Network Inter Method and the track process Method and the track process Second process and the track of the Resident and the track of the track of the Resident and the track of the track of the Method track of the track of the track of the Method track of the track of the track of the percent registration to the track of the percent registration.

Set to according to the set of th An extended price in a determined of the second of the sec







Processor – in larma di minar suscen and samering – ang-songes freg. We us they consist in pol fitteliti den Vielan en en ed. Progle free pinar Project, est mis ent affini i X kertitat nortee nave-good socces to thest through the H-th



One of the planes with the import paper the second entry into the closer states, second the intervention General Tensor for the Second entry (Control Is, Andrea reports to party the Perchal Town (m. 1). Knowledges and Cherris In. and The Sciences of Spacebook and Strammours. These prove formers on the basing order and second and shares and percent parts as interest. If Medical or one beam any parts and proves ingents as interest. Medical or one beam any parts and potent, if and there are registery also. engel 2014 Ande exan de Actor al tensor "Devel". The extra group is a Actor by group and many in All Annual in gran from the statement. Read many from Kick. Factores Histor (R. S. B Vanastanasa

Related Topics

Share this story over rate:





Boding cross, of the separation of the grant large is a heat in Figure of the more that the parts of vertices and the process, as a head on test to store. However, Johnson control and the hearts control was the test of anatoxic type called a start of the first hearts of the way. They anatoxic type called a start of the first heart of the start head of the start of type called a start of the start of the start head of the start head of the start of the start of the start of the start head of the start head of the start of the start of the start of the start head of the start head of the start head of the start of the start of the start of the start head of the start head of the start of the start of the start of the start head of the start head of the start of the start of the start of the start head of the start head of the start of the start of the start of the start head of the start head of the start of the start of the start of the start head of the start head of the start of the start of the start of the start head of the start head of the start of the start of the start of the start head of the start of the start of the start of the start head of the start of the start of the start of the start head the start of the start of the start of the start head the start of the start of the start of the start head the start of the start of the start of the start head the start of th . In seld to politics were some consistent A set production of the feed of product and the feed of the feed

How life experiency rees have slowed The definition of the second s

for the company memory.2010





In District, while a developed the District of the District

Is audonity to blams? No sources has been to see the constant devicts used. Bit he and these interesponded successful event order. In the and have entropy power's travely but super-travel. In a structure cost function are an welcoder, activity more and activity modifies interpreter of Defended Relativity by Mone of Degrees. *National Activity of the Structure personals on the Structure of the Structure Benerative* to In site highlighted what to value was "to weld" to the public weak to the NLS and apply a two which respective quarks of RT to compare public would have subcounce and and nucl-ited flats. The operations The way is called by proving local point for nonlinear place with the read of a strain of the second strain of the place of the second strain of the second



Secold-Parties may court form 19 strates and seeks free free little fields off diving our new siter challs RTC man is complete inter-Arming College & Theorem (ex. Educingus poet la Nama Patrati Cita verbilista Patra School Sourt, throats Elegeneer followers Agence Tell righteer Septier Couple 6 Methodal discount 7 Second the Day day Capition B Reservice and the Day day Capition B Reservice and The Segment of the B Reservice and The Segment of the B Reservice and the Segment of the Second Second

From Around the Web

The additional investment of the second seco And to so 1 incontinue was being matched one to the appropriate information was final panel or indiffer MMS leaded having been reasoned areas 2010 and estimation a total been interfed in a cost or in tan bend i tanebel in some incre konings Bhawanthamy in Pagand is durante (2 Mr. where, and 21 c. Mr. man. How do you improve your life expectancy? To don't areas to be hown. Thereas this will accounting space. To and age of a sing set in section is also repoted.

Dar 16 alexandra and was spid-Brook wythe broo

The government must act Galaxie 5 second dial second or Annue Sayles scal to Review was replice confider again (Nerrod access):



Chierd Prephanes and Annual and the interpretation of the of Prephanes and the interpretation of the interpretation of the theory of the repeated in party the formula Travertee. In Konstantion and Chernau Inc. and an This is secure of the called secul terminants. These cases factors curve to travery extensions writing contains and proved Lager south to SP Method to use the secure pair to map fact the information register also B Yes Creves

Related Topics

Share this story or car:



konterologi

seturae



Most Read 3 5

The other are have the consequences of endopsing undertailed tasks, and including the set over consect processed on on the off is solid tasks of another produces the point is also and, as conseting as includy supple or any time with reserve memory and tasks and as a solid set of the time with reserve memory.



 How long do we nailly care to bee?
 Limit is longer by reading the first pick.
 Limit versus larger to be set 54 beense How life expectancy rises have slowed

*2XHISPOR

From Around the Web

Denote a service on a large grade to service on the service of the

extract deterministication and the outer interview of a list of the outer interview of a list of the outer interview of a list of the outer interview of the out





 Han To Perror And Conference (margin: function and the solid system).
 Han Tomoral and Hang and How life expectancy rises have slower Comp Citize to Proteins Statistics proportionally to the formation 2000, S-Normal Arts base added with the tracement and David Lineth Constraints Stream 11 and an inclusion of the cooperation (and a labor to be 2010 in Transmit Linet and and color protein stream of 100 K. Ensend To 4 that leads of the series of the LK Edward Edwards Th, Bic operatory of bell interacted by one year only the years to carried on the series we many 2 3 years to meet Built to compare the one year carry 10 years for women and and for every set Ed-mony and With Interpret Mitt B. Michael, Annual reduced the treatment the tripping of XX, souther consoling graph is the commercy and testing constant to average graph is a ben't interpret to the second In Million, even we have a second operation of the product of the second product for a second three specific operations in the second operation of the second operation of the second operation oper

W2A0D0 W2A0D1 W2A0D2 News Opinion Sport Culture Lifestyle More-Alaola facas gui Sant-News Opinion Sport Colture Lifestyle More-Rolane Formatic India Ways Tarlah rojo: Stalkar Bill Tanke Fridal & Yr Son-News Ophinics System Collinare Lifestyle (Mater-Batter Aleman Hand, Shark Fragerunder #8) Batter Mater Structure (Kaces Christmas squeeze as inflation) Batter Mater Structure (Christmas squeeze as inflation) bite 2 4 4 6 contractions squeeze as inflation Business Business Projects lice UK faces Christmas squeeze as inflation hits 3.1%, outpacing wages - as it happened UK faces Christmas squeeze as inflation hits 3.1%, outpacing wages - as it happened UK faces Christmas squeeze as inflation hits 3.1%, outpacing wages - as it happened Business Inc. Proof, fact, and are facts and sec-in former rate elses Worth you Dates, Barl, sirilite form and rec-its fastest rate unter March 2017 Froit, fact, aictaic force and a Its fasters care since Narch 5 Heling hood prices mean Christmas filmer in Description and the same the christmas in the Shopper marks be same the christmas in Monty pound is cell caveling a over of Philippe Charts Pay is adopting boltain Back Natrix Carney ment explain in to Philip **Tarine** Of the lamped after fronts for graphice site s finnes will corr more In the other series of the ser Energined Energined Energined Shoppenson Weak points Charts Ray 1 Early Neret Earline off price per UK INFLATION HITS 2.1% 200 Connect Wearlant 5 0 5 10 10 10 10 10 6 9 10 10 10 10 201 079 ULIN UKINE LATION HITS 3.2% See Beating Childratin Tas Ni In Digensioned in channels years, in Betalary currentlying squeeze confirmer to bee. oras UK INFLATION HITS 32% ¹⁹⁷⁰ ¹ OR DEPARTMENT IN 5 (2005) Booking CR (influence has hits highest level inclusion at years, so relation score of hitsy spacese confluence to hits. The anomalo prime radio conclet V.1 in at the 12 nonlinear Forecastor, the highest level since Nard, 2012. f ¥ m ---meaners can of the lighter to extinue in both. The consume process many country () of an one country is to *Nonerrise*, the lighter of a dense before can be *Nonerrise*, the lighter of a dense behind the jump, some dang to the other for that and backets, along order counterband and onload goods and here can also a game. Jump -mentero The converse prior information to set your the conversion Knowledge to higher from the Study Set you Ar base converse hyper from the set of comp, even instantistic Structure from the set of the set of the set of the set of the Structure set of the set o Ar faces were in https/finite/bland the unit, according to in White for Patienal Structure along with recommend and call and goods and arrive socials a computer genes. 23* (20wr 200 - une arrit and the second articles shall be also also grants. Web/Al-K-summage with memory by neural 2.1% to 2.2% an assess methods, the second-second-second for early filling that mage is the second second second second second second second second filling with a filling that the complete second second second between connected or 2 ming themsel to second as the owners of the second second second. provide all arrays and a secondary taxos. With K range on address that the second s la 315 8 doe 200 Careyoder S Actor With the sample only many, by around 2, 2% to 2.3% in cases more the, the more characteristic are will effort a hilling out a same 0 125-1 Core of The off classifiers, Each of Explant governer Hack Cannoy Levent, a few off classifier (Help Exempted is explain only inflation test) for a few to describe a Help Exempted is explain only inflation test) 0.25-B 04-Mars in Johns' Marcadillee f = 5: 0 220 220 View UK inflation may by LUS in the UK months in November, the November into March 2012, while average weeks excluded to L228 Look and L228 Look and L228 Via . Via UK inflation note by 3.15 in the 32 months in Kovember, the higher level sizer March 2012 wildo reage weaky earlings mode 7.255 1 25-220 Har court and Conversion that -Elkine): Thei Sign Fictor 0 75-Ann 110m 207 B Olto capensive Outstma diment as food price inflation ware 236 200 Point record of Tana. 200 200 - Standard and a spec-solation of the Dille: MO THE MERIC IN THESE IN THE MERIC IN THE IN 122.5

W2A1D0

W2A1D1

W2A1D2





Appendix A5: Debrief Page

Experiment - Bias Perception in Online News

Thank You!

For participating in this crowdsourced experiment

Debrief

This research is being conducted to better understand the impact of professional design on the perception of Bias in online News. Consequently, it was necessary to modify existing news articles to increase or decrease the amount of Bias in each. Author names were also changed.

As such, none of the information presented here can be counted as an accurate, fair or balanced account of any of the topics covered.

If you are interested in further information or news on any of the topics depicted in this experiment I would encourage you to seek out genuine news on each topic from multiple reputable news sources.

Each of the websites used in the experiment were copies of the websites of existing reputable news agencies. The news agencies are not involved in this research and the articles in this experiment should not be attributed to them.

Return to Prolific Academic

If you would like to contact the researcher leading this experiment regarding any aspect of it please email:

Cian Flynn

© Cian Flynn, Trinity College Dublin 2018. Email: Cian Flynn.
Appendix A6: Background Results

This appendix contains visualisations of links between social and economic data provided by participants, and the bias ratings they provided.

The group codes correspond to those detailed in Appendix A2. The results are to be interpreted as per Section 6.1.

A6.1 Political Leaning vs Reported Bias, Daily Mail



Bias Reported							
Political Leaning	Mean	N	Std. Deviation				
0	33.61	28	27.341				
1	27.27	15	25.401				
2	37.04	25	36.316				
3	45.44	25	30.053				
4	29.86	7	21.950				
5	29.00	11	30.760				
6	50.00	11	37.701				
8	4.00	2	5.657				
Total	36.27	124	30.934				

Report

X²(7) = 8.533, *p*-value = 0.288

A6.2 Political Leaning vs Reported Bias, BBC





Report

Bias Reported								
Political Leaning	Mean	N	Std. Deviation					
0	32.89	28	25.382					
1	40.87	15	31.514					
2	33.08	25	31.070					
3	42.52	25	30.817					
4	33.86	7	31.445					
5	39.45	11	33.512					
6	52.64	11	33.494					
8	1.00	2	1.414					
Total	37.71	124	30.289					

A6.3 Political Leaning vs Reported Bias, Daily Mail



Report

Bias Reported							
Political Leaning	Mean	N	Std. Deviation				
0	30.93	28	29.747				
1	27.13	15	25.809				
2	32.72	25	29.457				
3	31.28	25	30.066				
4	16.86	7	5.146				
5	14.09	11	20.945				
6	46.73	11	40.856				
8	29.00	2	41.012				
Total	29.98	124	29.371				

X²(7) = 6.175, *p*-value = 0.519

A6.4 Occupation vs Reported Bias



Occupation	Mean	N	Std. Deviation
1	27.21	33	27.810
2	27.42	60	22.892
3	38.58	24	32.931
4	41.49	117	34.456
5	32.33	39	29.811
6	32.69	51	30.020
7	29.50	6	29.508
8	30.83	12	26.208
9	46.67	6	32.432
10	33.92	24	24.746
Total	34.66	372	30.310

Bias Reported

X²(9) = 10.240, *p*-value = 0.331

Appendix A7: Instruction Tasks

Instruction Task 1

ant 🗿 Roostan Spil thejournal. iction Task 1 e presented with a ser a hist fulfile interaction Please road each news article and use the Ulder at t screen to rate, fram 0 to 100, have blased the article Greens co-leade admitting to we -----MOAL T App Store NATIONAL 45.2% 2.2
Use of the second se Ireland's undocumented ingrantic 'Our flat was broken into. We couldn't report it to the this Column being an explanat FORA THE CO Trans enable importantia the Matrix S Galway Intelling great Theory Koody antiously ill in herp D 10 Decek E Roveske that will Rei a 10 0 2 m 100 0 0 -

Instruction Task 2 My N A 6 (?) 🦛 R.R. The second secon thejournalie Macron abandons plan to make his wife France's first-ever First Lady State & A Pa SHI4 App Store J arment 196 1 TE Con La Martin Junes Coherri and Hidy will slong by manakon La a anadago Uniday, Maliy consen and Thema Magna hashi an to a yola prior activity of and his in factor of the theory of the one dimensioned the state of the theory of the theory of the second History (and History) tasks to the the second History (activity) 100 0 0 100

Instruction Task 1 Popup Instructions

Instruction Task 1

You will be presented with a series of news webpages, each containing a single news article.

Please read each news article and use the Slider at the bottom of the screen to rate, from 0 to 100, how Biased the article is.

0 being Unbiased and 100 being extremely Biased

Bias in news articles is defined as:

"Deliberate or accidental slant by the journalist, editor or publication to distort reality."

I Understand

×

Instruction Task 2 Popup Instructions



Appendix A8: D1 and D2 Graphics





This graphic massively exaggerates the difference between CPI inflation and wage inflation, while also being relatively difficult to extract any precise information from.





Price of bitcoin in US Dollars over three months, with overall trend in red. Data source: coindesk.com

This graphic adds a bold red downward trendline to the price of bitcoin graph, and arrow showing the difference from the start of the graph. This trendline is ignores any nuance in the graphic, by being bolder and more attention grabbing than the line itself, and bolsters the narrative of an overall sharp decline in price.

W0A1D1 Graphic



W0A2D2 Graphic



This graphic presents the information differently to the D1 graphic, giving the measure in change of life expectancy rather than just life expectancy. The graphic also appears to be incomplete.





This graphic massively exaggerates the difference between CPI inflation and wage inflation, while also being relatively difficult to extract any precise information from.





W1A1D2 Graphic

Data source: coindesk.com

This graphic adds a bold red downward trendline over the actual price of bitcoin, and arrow showing the difference from the start of the graph. This trendline is ignores any nuance in the graphic, by being bolder and more attention grabbing than the line itself, and bolsters the narrative of an overall sharp decline in price.





This graphic focusses on a different detail picked out in the article than the original graphic, although it is from the same cited data source. The article's narrative suggests a link between decline rate of life expectancy increase with decline healthcare spend in an apparent politic point. This graphic aims to bolster that narrative.

W1A2D2 Graphic





This graphic massively exaggerates the difference between CPI inflation and wage inflation, while also being relatively difficult to extract any precise information from.



W2A1D2 Graphic



Price of bitcoin in US Dollars over three months, with overall trend in red. Data source: coindesk.com

This graphic adds a bold red downward trendline to the actual price of bitcoin, and arrow showing the difference from the start of the graph. This trendline is ignores any nuance in the graphic, by being bolder and more attention grabbing than the line itself, and bolsters the narrative of an overall sharp decline in price.



W2A2D2 Graphic



This graphic presents the information differently to the D1 graphic, giving the measure in change of life expectancy rather than just life expectancy. The graphic also appears to be incomplete.

	Р										
DF	0.995	0.975	0.20	0.10	0.05	0.025	0.02	0.01	0.005	0.002	0.001
1	0.0000393	0.000982	1.642	2.706	3.841	5.024	5.412	6.635	7.879	9.550	10.828
2	0.0100	0.0506	3.219	4.605	5.991	7.378	7.824	9.210	10.597	12.429	13.816
3	0.0717	0.216	4.642	6.251	7.815	9.348	9.837	11.345	12.838	14.796	16.266
4	0.207	0.484	5.989	7.779	9.488	11.143	11.668	13.277	14.860	16.924	18.467
5	0.412	0.831	7.289	9.236	11.070	12.833	13.388	15.086	16.750	18.907	20.515
6	0.676	1.237	8.558	10.645	12.592	14.449	15.033	16.812	18.548	20.791	22.458
7	0.989	1.690	9.803	12.017	14.067	16.013	16.622	18.475	20.278	22.601	24.322
8	1.344	2.180	11.030	13.362	15.507	17.535	18.168	20.090	21.955	24.352	26.124
9	1.735	2.700	12.242	14.684	16.919	19.023	19.679	21.666	23.589	26.056	27.877
10	2.156	3.247	13.442	15.987	18.307	20.483	21.161	23.209	25.188	27.722	29.588
11	2.603	3.816	14.631	17.275	19.675	21.920	22.618	24.725	26.757	29.354	31.264
12	3.074	4.404	15.812	18.549	21.026	23.337	24.054	26.217	28.300	30.957	32.909
13	3.565	5.009	16.985	19.812	22.362	24.736	25.472	27.688	29.819	32.535	34.528
14	4.075	5.629	18.151	21.064	23.685	26.119	26.873	29.141	31.319	34.091	36.123
15	4.601	6.262	19.311	22.307	24.996	27.488	28.259	30.578	32.801	35.628	37.697
16	5.142	6.908	20.465	23.542	26.296	28.845	29.633	32.000	34.267	37.146	39.252
17	5.697	7.564	21.615	24.769	27.587	30.191	30.995	33.409	35.718	38.648	40.790
18	6.265	8.231	22.760	25.989	28.869	31.526	32.346	34.805	37.156	40.136	42.312
19	6.844	8.907	23.900	27.204	30.144	32.852	33.687	36.191	38.582	41.610	43.820
20	7.434	9.591	25.038	28.412	31.410	34.170	35.020	37.566	39.997	43.072	45.315

Appendix A9: Chi-squared Tables

Source: https://www.medcalc.org/manual/chi-square-table.php